Discovering the Relationship Between Generalization and Uncertainty by Incorporating Complexity of Classification

Xi-Zhao Wang, Fellow, IEEE, Ran Wang, Member, IEEE, and Chen Xu

Abstract—The generalization ability of a classifier learned from ² a training set is usually dependent on the classifier's uncertainty, 3 which is often described by the fuzziness of the classifier's outputs 4 on the training set. Since the exact dependency relation between 5 generalization and uncertainty of a classifier is quite compli-6 cated, it is difficult to clearly or explicitly express this relation in 7 general. This paper shows a specific study on this relation from 8 the viewpoint of complexity of classification by choosing extreme 9 learning machines as the classification algorithms. It concludes 10 that the generalization ability of a classifier is statistically becom-¹¹ ing better with the increase of uncertainty when the complexity 12 of the classification problem is relatively high, and the general-13 ization ability is statistically becoming worse with the increase 14 of uncertainty when the complexity is relatively low. This paper 15 tries to provide some useful guidelines for improving the gener-16 alization ability of classifiers by adjusting uncertainty based on 17 the problem complexity.

Index Terms—Complexity of classification, extreme learning
 machine, generalization, uncertainty.

20

I. INTRODUCTION

²¹ C LASSIFICATION problem, as the central part in the ²² C fields of pattern recognition and data mining, refers ²³ to a task of assigning objects to one of several predefined ²⁴ class labels. Given a set of objects, the mathematical ²⁵ model of classification problem is a discrete-valued func-²⁶ tion that maps each object to a class label. Usually, the ²⁷ process of determining the discrete-valued function from a

Manuscript received July 4, 2016; revised November 28, 2016; accepted January 11, 2017. This work was supported in part by the National Natural Science Foundation of China under Grant 61402460, Grant 61472257, Grant 61170040, and Grant 71371063, in part by the Basic Research Project of Knowledge Innovation Program in Shenzhen under Grant JCYJ20150324140036825, in part by the Guangdong Provincial Science and Technology Plan Project under Grant 2013B040403005, and in part by the HD Video Research and Development Platform for Intelligent Analysis and Processing in Guangdong Engineering Technology Research Centre of Colleges and Universities under Grant GCZX-A1409. (*Corresponding author: Ran Wang.*)

X.-Z. Wang is with the College of Computer Science and Software Engineering, Shenzhen University, Shenzhen 518060, China (e-mail: xizhaowang@ieee.org).

R. Wang and C. Xu are with the College of Mathematics and Statistics, Shenzhen University, Shenzhen 518060, China (e-mail: wangran@szu.edu.cn; xuchen@szu.edu.cn).

Color versions of one or more of the figures in this paper are available online at http://ieeexplore.ieee.org.

Digital Object Identifier 10.1109/TCYB.2017.2653223

training set is called learning while the process of using ²⁸ the determined function to classify a new object is called ²⁹ reasoning [1]–[5]. ³⁰

For a classification problem with c classes, the reason-31 ing result is generally a *c*-dimensional vector. According to 32 the output forms of the reasoning process, existing learning 33 algorithms can be classified into two categories. In one cate-34 gory, the *c*-dimensional output vector contains one component $_{35}$ of value 1 and other components of value 0. In this situa-36 tion, the class label corresponding to the component 1 will 37 be the reasoning result. This kind of algorithms are known 38 as crisp-output algorithms, such as traditional support vector 39 machine (SVM) [6]–[10], decision tree (DT) [11], [12], etc. 40 In the other category, the c-dimensional output vector con- $_{41}$ tains components of real values within the interval [0, 1]. In 42 this situation, the class label corresponding to the maximum 43 component will be the reasoning result. If the maximum is 44 attained at more than one component, a special strategy will be 45 designed to determine the final result. This kind of algorithms 46 are acknowledged as uncertain-output algorithms, such as 47 k-nearest neighbor [2], Bayesian probability model [2], back-48 propagation (BP) methods for training feed-forward neural 49 networks [13]-[16], etc. 50

Obviously, crisp-output algorithms are special cases of 51 uncertain-output algorithms. If an algorithm belongs to the 52 crisp category, then it belongs to the uncertain category, 53 however, it is not true conversely. Most crisp-output algo-54 rithms can be extended to uncertain-output algorithms, such 55 as fuzzy SVM [17], fuzzy DT [18], etc. In this paper, we 56 will intensively investigate the uncertain-output algorithms, 57 which highlight the argument that uncertainty does exist in 58 the learning and reasoning processes. 59

On the other hand, generalization of a classifier is defined ⁶⁰ as the rate of the correctly classified objects that are not in ⁶¹ the training set. It is the most important index for evaluating a ⁶² classification algorithm since the ultimate goal for developing ⁶³ a classification model is to achieve high prediction accuracy ⁶⁴ on unseen cases. Usually, the generalization of a classifier ⁶⁵ depends on multiple factors. ⁶⁶

- The mathematical model, which has a direct impact on both the training accuracy and testing accuracy.
- 2) The algorithm for training the model parameters, which ⁶⁹ is sensitive to the prediction results. ⁷⁰

2168-2267 © 2017 IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See http://www.ieee.org/publications_standards/publications/rights/index.html for more information.

129

132

71 3) *The data distribution:* In supervised learning, there is
72 a fundamental assumption that the training data has
73 the same distribution as the testing data. The learning
74 scheme that does not follow this fundamental assump-

tion is referred to as transfer learning [19], which is out

⁷⁶ of the scope of this paper.

77 Many research efforts have been made to improve the gen-78 eralization of a classifier by considering different factors. this paper, we consider a particular model parameter, 79 In 80 i.e., the uncertainty of the classifier's outputs, which has been ⁸¹ proven in [20] to have a close relationship with the gen-82 eralization of classifier. It has been shown in [20] that the ⁸³ uncertainty of the classifier's outputs has a close relationship 84 with the generalization capability. However, this relation-85 ship is difficult to express explicitly for general cases. In 86 order to further investigate this relationship, in this paper, 87 we take into account a new index, i.e., complexity of clas-⁸⁸ sification, which can be measured in different ways [21]. To 89 the best of our knowledge, this paper makes a first attempt investigate the relationship between generalization and 90 to ⁹¹ uncertainty of a classifier by incorporating the complexity of 92 classification.

In addition, choosing an appropriate classification algorithm 93 also an important issue to conduct this research. It is note-94 is 95 worthy that any uncertain-output algorithm can be used to ⁹⁶ study the relationship between generalization and uncertainty. 97 As the commonly used classification model for various prac-98 tical problems, feed-forward neural networks will be adopted. ⁹⁹ The most notable algorithm to train a feed-forward neural network is BP. Although it has been proved in [15] and [16] 100 101 that BP network has the ability to approximate any contin-102 uous function with arbitrary precision, it is often criticized 103 to have the problems of slow convergence speed and local 104 minima. In order to overcome these deficiencies, extreme lean-105 ing machine (ELM) has been proposed as a new training 106 algorithm for single-hidden layer feed-forward neural net-107 work (SLFN) [22]. Differentiating from BP that iteratively 108 tunes the weight parameters by gradient descent technique, 109 ELM randomly chooses the weight parameters between input 110 and hidden layers and analytically solves the weight parame-111 ters between hidden and output layers through Moore–Penrose ¹¹² generalized inverse [44]–[48]. Due to the extremely fast train-¹¹³ ing speed and good prediction performance, ELM has been 114 investigated intensively and extensively in the machine learn-115 ing and data mining communities [23]–[26]. Based on the 116 aforementioned advantages, we will adopt ELM as the classi-¹¹⁷ fication algorithm in this paper. The major theoretical issues 118 of ELM can be found in [27] and [28], and the applications ¹¹⁹ of ELM to different areas, such as sparse representation can 120 be found in [29] and [30].

The rest of this paper is organized as follows. Section II reviews ELMs. Section III introduces the dependency relation between generalization and uncertainty of classifiers. Section IV discusses the complexity of classification problems. Section V analyzes the relationship between generalization and uncertainty by incorporating a complexity index. Experiments review of the section VI. Finally, conclusions are given in Section VII.

II. EXTREME LEARNING MACHINE

This section will introduce ELM, which is a noniterative ¹³⁰ training algorithm for SLFNs. ¹³¹

A. Training of ELM

A standard SLFN for classification is a discrete function ¹³³ mapping samples to class labels. Given a training set that ¹³⁴ contains *N* arbitrarily distinct samples $\mathbb{X} = \{(\mathbf{x}_i, \mathbf{t}_i)\}_{i=1}^N \subset ^{135}$ $\mathcal{R}^n \times \{0, 1\}^c$, where $\mathbf{x}_i = [x_{i1}, x_{i2}, \dots, x_{in}]$ is the *i*th training ¹³⁶ sample, $\mathbf{t}_i = [t_{i1}, t_{i2}, \dots, t_{ic}]$ is the label vector of \mathbf{x}_i , *n* is ¹³⁷ the number of features, and *c* is the number of classes. An ¹³⁸ SLFN with \tilde{N} hidden nodes and activation function $g(\mathbf{x})$ can ¹³⁹ be expressed as

$$\sum_{j=1}^{\widetilde{N}} \beta_j g(\mathbf{w}_j \cdot \mathbf{x}_i + b_j) = \mathbf{t}_i, \ i = 1, 2, \dots, N$$
(1) 141

where $\mathbf{w}_j = [w_{j1}, w_{j2}, \dots, w_{jn}]$ is the weight linking the input 142 nodes to the *j*th hidden node, b_j is the bias of the *j*th hidden 143 node, β_j is the weight linking the *j*th hidden node to the output nodes, and sigmoid function $g(x) = (1/[1 + \exp(-x)])$ is 145 selected as the activation function. 146

In ELMs, the input weights \mathbf{w}_j and biases b_j are randomly 147 chosen, and the learning can be formulated as a minimum 148 optimization problem with a regularized term 149

$$\min_{\beta} \left\{ ||\mathbf{T} - \mathbf{H}\beta||_{2}^{2} + \mu ||\beta||_{2}^{2} \right\}, \ \mu > 0$$
 (2) 150

where **H** is the hidden layer output matrix denoted as

$$\mathbf{H}(\mathbf{w}_1, \mathbf{w}_2, \dots, \mathbf{w}_{\widetilde{N}}, b_1, b_2, \dots, b_{\widetilde{N}}, \mathbf{x}_1, \mathbf{x}_2, \dots, \mathbf{x}_N)$$

$$= \begin{bmatrix} g(\mathbf{w}_1 \cdot \mathbf{x}_1 + b_1) & \cdots & g(\mathbf{w}_{\widetilde{N}} \cdot \mathbf{x}_1 + b_1) \\ \vdots & \vdots & \vdots \end{bmatrix}$$
(3)

$$\begin{bmatrix} \vdots & \ddots & \vdots \\ g(\mathbf{w}_1 \cdot \mathbf{x}_N + b_1) & \cdots & g(\mathbf{w}_{\widetilde{N}} \cdot \mathbf{x}_N + b_{\widetilde{N}}) \end{bmatrix}_{N \times \widetilde{N}}$$

and **T** is the label matrix denoted as

151

$$\mathbf{T} = \begin{bmatrix} \mathbf{t}_1 \\ \vdots \\ \mathbf{t}_N \end{bmatrix}_{N \times c}$$
(4) 155

The optimal estimation of output weights β^* can be formulated as a regularized least square problem

$$\boldsymbol{\beta}_{\widetilde{N}\times c}^* = \left(\mathbf{H}^{\mathrm{T}}\mathbf{H} + \boldsymbol{\mu}\mathbf{I}\right)^{-1}\mathbf{H}^{\mathrm{T}}\mathbf{T}$$
(5) 158

where I is the identity matrix of suitable dimension and μ is the regularizing factor.

To this end, all the parameters $\{\mathbf{w}, b, \beta\}$ in ELM have been ¹⁶¹ fixed, and the training process is finished. ¹⁶²

ELMs have been proved to have the universal approximation capabilities [31] although the training process does not include any iteration. Under the assumption of smoothness of the underlying function, the universal approximation capability of ELMs can be guaranteed by providing a sufficiently large number of hidden nodes with certain range of **w** and *b*. 168

In comparison with BP algorithm, ELMs have a much faster 169 training speed due to the noniterative mechanism. References 170 show that ELMs can finish the training process thousands of 171 times faster than BP in some scenarios, at the same time, an 172 ¹⁷³ acceptable learning accuracy is kept. The advantages and dis-¹⁷⁴ advantages of ELMs are listed in Appendix A. Furthermore, ¹⁷⁵ one can find many improved versions for ELMs. The com-¹⁷⁶ putation of weights between hidden and output layers can ¹⁷⁷ be improved through an optimization algorithm given by ¹⁷⁸ Deng *et al.* [32] in order to avoid over-fitting. Rong *et al.* [33] ¹⁷⁹ offered a pruned ELM in which the corresponding nodes ¹⁸⁰ can be removed according to the information gain to reduce ¹⁸¹ the correlation among classes in a large network structure. ¹⁸² Feng *et al.* [34] proposed an EM-ELM in which the weights ¹⁸³ are not updated when a node is added, and the algorithm ¹⁸⁴ can update the weights and adjust the network at the same ¹⁸⁵ time. Furthermore, it is found that ELMs can online deal with ¹⁸⁶ sequential data successfully [35].

187 B. Generalized Inverse and Normal Equations

In ELMs, the weights between hidden and output layers are calculated by the generalized inverse [36]. We briefly review some connections between the generalized inverse and the normal equations. Originally, the training of ELMs contains two parts. The first is to randomly assign values in a specified interval to the weights between the input and hidden layers while the second is to determine the weights between the hidden and output layers by computing the generalized inverse of the matrix **H** as $\beta^* = \mathbf{H}^{\dagger}\mathbf{T}$. It is the minimum norm and minimum least square solution of the system of linear matrix equations $\mathbf{H}\beta = \mathbf{T}$. It is easy to prove that, if the matrix **H** is of full-rank, the solution of normal equation $\mathbf{H}^{T}\mathbf{H}\beta = \mathbf{H}^{T}\mathbf{T}$ is identical to $\beta^* = \mathbf{H}^{\dagger}\mathbf{T}$.

Noting that in Section II-A, the training process of ELMs is 201 written as $\beta^* = (\mathbf{H}^T \mathbf{H} + \mu \mathbf{I})^{-1} \mathbf{H}^T \mathbf{T}$, where μ is a regularizing 202 factor. This formula is identical to $\beta^* = \mathbf{H}^{\dagger}\mathbf{T}$ if the regulariz-203 ing factor takes value zero. It is proven in [24] that the matrix 204 is of full-rank with probability 1, and therefore, we can say Η 205 ²⁰⁶ that the solution of normal equation $\mathbf{H}^{\mathrm{T}}\mathbf{H}\beta = \mathbf{H}^{\mathrm{T}}\mathbf{T}$ is available with probability 1. In fact, the regularizing factor, which 207 208 makes the solved weights as small as possible, has the effect 209 to become the matrix **H** full of rank.

Practically the number of rows is much larger than the number of columns for an input data matrix. It implies that the transformation from computing $\beta^* = \mathbf{H}^{\dagger}\mathbf{T}$ to solving the normal system of linear matrix equations $\mathbf{H}^{T}\mathbf{H}\beta = \mathbf{H}^{T}\mathbf{T}$ and save much computational load, since the order of \mathbf{H} is $N \times \tilde{N}$ but the order of $\mathbf{H}^{T}\mathbf{T}$ is $\tilde{N} \times c$, where N is the number of input samples, \tilde{N} is the number of hidden layer nodes, and c is the number of classes. A lot of numerical experiments have confirmed this saving of computational load.

220 III. DEPENDENCY RELATION BETWEEN 221 GENERALIZATION AND UNCERTAINTY 222 OF CLASSIFIERS

In this section, we will introduce the generalization and uncertainty of a classifier. The dependency relation between generalization and uncertainty is then discussed.

A. Generalization and Uncertainty

Generally speaking, the purpose of learning is to acquire ²²⁷ the knowledge hidden in the data. Knowledge representation, ²²⁸ which has been well acknowledged as a bottle-neck problem ²²⁹ in machine learning and artificial intelligence for many years, ²³⁰ does not have a general definition but has many specific forms. ²³¹ A mathematical model, such as a set of IF-THEN rules or a ²³² neural network learned from a training set, can be regarded ²³³ as a typical form of knowledge representation. The ability ²³⁴ or performance of the learned model to predict unseen cases ²³⁵ (which are not within the training set) is called generalization. ²³⁶

Let S be a finite space of samples, $F(\mathbf{x})$ be a discrete-valued ²³⁷ function defined on S, and \mathbb{X} be a subset of S. Based on values ²³⁸ of $F(\mathbf{x})$ in \mathbb{X} , an estimator function $f(\mathbf{x})$ defined on S is given ²³⁹ by using a training algorithm. The discrete-valued function ²⁴⁰ $f(\mathbf{x})$ has the same value range as $F(\mathbf{x})$. Usually we call $f(\mathbf{x})$ ²⁴¹ as a classifier trained by the algorithm on \mathbb{X} . ²⁴²

Definition 1: The generalization of classifier $f(\mathbf{x})$ is 243 defined as 244

$$G(f) = \frac{|\{\mathbf{x} : \mathbf{x} \in \mathcal{S} - \mathbb{X}, F(\mathbf{x}) = f(\mathbf{x})\}|}{|\mathcal{S} - \mathbb{X}|}$$
(6) 245

where | | denotes the number of elements in a set.

Generalization is the most important index of evaluating ²⁴⁷ a learned model. From mathematical viewpoint, the task of ²⁴⁸ learning is to find a function $f(\mathbf{x})$ through a training set ²⁴⁹ $\mathbb{X} = \{(\mathbf{x}_i, \mathbf{t}_i)\}_{i=1}^N \subset \mathcal{R}^n \times \{0, 1\}^c$ such that $f(\mathbf{x})$ can well ²⁵⁰ approximate the objective function $F(\mathbf{x})$ both at training cases ²⁵¹ and unseen cases. The difference between $F(\mathbf{x})$ and $f(\mathbf{x})$ is ²⁵² called generalization error, which can be measured from different angles. One method is to estimate an upper bound for it, the ²⁵⁴ other is to compute $R = \int_{\mathcal{S}} [F(\mathbf{x}) - f(\mathbf{x})]^2 p(\mathbf{x}) d\mathbf{x}$, where $p(\mathbf{x})$ ²⁵⁵ is the probability density function of input \mathbf{x} . Experimentally, ²⁵⁶ the generalization can be measured by the prediction accuracy ²⁵⁷ of the classifier on a testing set.

Multiple factors have critical impacts on the generalization 259 of a classifier. 260

- Model Selection: It is hard to select the most appropriate ²⁶¹ model for a given classification task. When the training ²⁶² data is fixed, the generalizations of two models might ²⁶³ be quite different. This is due to the data distribution, ²⁶⁴ i.e., a model suitable for one type of data may not be ²⁶⁵ appropriate for another type of data. ²⁶⁶
- Training Algorithm: When a model is fixed, the subsequent work is to train the model parameters based on a training set. A model with a set of trained parameters 269 has the generalization quite different from the model 270 with another set of trained parameters. 271
- 3) Representatives of Training Data: Since both the objec- 272 tive function and its approximating function are defined 273 on a space S, one problem is that the training set 274 X should be a reasonable sampling of the space S, 275 which directly relates to the fundamental assumption of 276 machine learning that the training set has an identical 277 distribution as the testing set has. 278
- 4) *Model Knowledge Parameters:* Different from the 279 parameters inside the model that are acquired directly 280

from the training process, model knowledge parame-281 ters do not explicitly appear in the model, which are 282 usually evaluated after the training process. For exam-283 ple, the uncertainty of classifier's outputs is a typical 284 model knowledge parameter. The relationship between 285 generalization and uncertainty of a classifier is initially 286 demonstrated in [20]. This paper will conduct further 287 studies on this relationship through incorporating a new 288 index, i.e., complexity of classification. 289

290 B. Fuzziness of Classifier's Outputs

In this paper, we use fuzziness to depict the uncertainty 291 292 of a classifier's outputs. The term "fuzziness," in conjunc-293 tion with the concept of fuzzy set, was first mentioned by ²⁹⁴ Zadeh [37]. He also generalized a probability measure of 295 events that cannot be described by sharply defined collection points, and suggested using entropy in information the-296 Of ry to interpret the uncertainty associated with a fuzzy event. 297 01 De Luca and Termini [38] for the first time clearly proposed 298 ²⁹⁹ three properties that a fuzziness measure should satisfy. The term fuzziness can be interchangeable with "ambiguity" in 300 some scenarios. Klir et al. [39], [40] stated that fuzziness and 301 ³⁰² ambiguity gave two cognitive uncertainty measures.

As stated in [41], the fuzziness of a fuzzy set μ can be measured by a mapping $E(\mu):F(S) \rightarrow [0, \infty]$ where F(S)of denotes the space of all fuzzy sets defined on S, satisfying the following axioms.

1) $E(\mu) = 0$ if and only if μ is a crisp set.

2) $E(\mu)$ attains its maximum value if and only if $\forall \mathbf{x} \in \mathcal{S}: \mu(\mathbf{x}) = 0.5$.

310 3) If $\mu \leq_s \sigma$, then $E(\mu) \geq E(\sigma)$, where \leq_s is defined as

$$\begin{array}{ll} \underset{11}{\underset{12}{\text{min}}} & \mu \leq_s \sigma \Leftrightarrow \min(0.5, \mu(\mathbf{x})) \geq \min(0.5, \sigma(\mathbf{x})) \\ \underset{12}{\underset{12}{\text{max}}} & \max(0.5, \mu(\mathbf{x})) \leq \max(0.5, \sigma(\mathbf{x})) \end{array}$$

313 4)
$$E(\mu) = E(\mu')$$
 when $\forall \mathbf{x} \in \mathcal{S}: \mu'(\mathbf{x}) = 1 - \mu(\mathbf{x})$

314 5) $E(\mu \cup \sigma) + E(\mu \cap \sigma) = E(\mu) + E(\sigma).$

³¹⁵ Based on these axioms, we further introduce the following ³¹⁶ definition.

Definition 2 [32]: Let $B = \{\mu_1, \mu_2, \dots, \mu_m\}$ be a fuzzy site set, the fuzziness of B can be defined as

³¹⁹
$$E(B) = -\frac{1}{m} \sum_{i=1}^{m} \left(\mu_i \log \mu_i + (1 - \mu_i) \log(1 - \mu_i) \right).$$
(7)

 $_{320}$ It is easy to verify that formula (7) indeed satisfies $_{321}$ axioms 1–5.

Given a set of samples $\mathbb{X} = \{(\mathbf{x}_i, \mathbf{t}_i)\}_{i=1}^N \subset \mathcal{R}^n \times \{0, 1\}^c$ and a well-trained classifier, a membership matrix $\mathbf{U} = [\mu_{ij}]$ are can be obtained by matching each sample to the classifier, where $\mu_{ij} = \mu_j(\mathbf{x}_i)$ denotes the membership degree of the the sample belonging to the *j*th class, where i = 1, 2, ..., Nare and j = 1, 2, ..., c. It is worth noting that each output vector may not be a probability distribution, i.e., $\mu_{ij} \in [0, 1]$, and the equality $\sum_{i=1}^{c} \mu_{ij} = 1$ does not necessarily hold. Based on Definition 2, the fuzziness of the classifier's 330 outputs for the *i*th sample can be expressed as 331

$$E(\mu_i) = -\frac{1}{c} \sum_{j=1}^{c} \left(\mu_{ij} \log \mu_{ij} + \left(1 - \mu_{ij}\right) \log(1 - \mu_{ij}) \right).$$
(8) 332

Having the above preliminaries, in the following, we propose a new concept to describe the fuzziness of a classifier's outputs on the entire training set.

Definition 3 (Fuzziness of a Classifier's Outputs): Suppose ³³⁶ that a classifier is trained from training set X. Without loss ³³⁷ of generality, X is assumed to be a sufficient sampling of ³³⁸ the entire sample space. Let $\mathbf{U} = [\mu_{ij}]_{c \times N}$ be the membership ³³⁹ matrix given by matching each training sample to the classifier, ³⁴⁰ where *c* is the number of classes and *N* is the number of ³⁴¹ samples. Then the fuzziness of the classifier's outputs can be ³⁴² defined as ³⁴³

$$E(\mathbf{U}) = -\frac{1}{cN} \sum_{i=1}^{N} \sum_{j=1}^{c} (\mu_{ij} \log \mu_{ij} + (1 - \mu_{ij}) \log(1 - \mu_{ij})).$$
⁽⁹⁾ 344

It is noted that Definition 3 uses the fuzziness of the classifier's outputs on the training set. In a more rigorous manner, it should be defined as the fuzziness of the classifier on the whole space. Unfortunately, the fuzziness of the classifier on unseen samples is unknown. According to the fundamental assumption of supervised learning that the training set is a reasonable and sufficient sampling of the entire sample space, we can use the classifier's fuzziness on the training set to approximately replace the classifier's fuzziness on the entire sample space. 354

C. Relationship Between Generalization and Fuzziness

355

Previous study [20] shows that the classifier with higher ³⁵⁶ fuzziness of outputs has a better generalization for com-³⁵⁷ plex boundary problems when the training accuracy attains ³⁵⁸ a predefined threshold. Furthermore, it demonstrates that the ³⁵⁹ outputs of boundary samples have higher fuzziness, and ³⁶⁰ samples with higher fuzziness exhibit higher risk of misclas-³⁶¹ sification. By separating samples with high fuzziness from ³⁶² samples with low fuzziness, a divide-and-conquer learning ³⁶³ algorithm based on fuzziness categorization was proposed ³⁶⁴ in [41]. It shows that the category of sample with low or high ³⁶⁵ fuzziness plays a critical role for performance improvement. ³⁶⁶ Although these studies confirm that a relationship between ³⁶⁷ fuzziness and generalization of a classifier indeed exists, it is ³⁶⁸ difficult to explicitly express this relationship in general. ³⁶⁹

In the following, we make an investigation on data set *Spam*, ³⁷⁰ which is a binary classification data set selected from UCI ³⁷¹ machine learning repository. This data set contains 4601 sam- ³⁷² ples with 57 features. We randomly split it into two parts, ³⁷³ i.e., 70% for training and 30% for testing. ELM is used to ³⁷⁴ construct a classifier, which generates four indexes, i.e., train- ³⁷⁵ ing accuracy, testing accuracy, training fuzziness, and testing ³⁷⁶ four indexes are recorded for each repetition. ³⁷⁸

We make a statistical analysis for the 100 results. First, we 379 split the interval between the minimum and maximum fuzzi- 380 ness values into ten parts with equal length and generate ten 381



Fig. 1. Dependency relation between fuzziness and accuracy for *Spam*. (a) Histogram of training fuzziness. (b) Training accuracy. (c) Histogram of testing fuzziness. (d) Testing accuracy.

³⁸² levels of fuzziness. For instance, the minimum and maximum ³⁸³ fuzziness values for testing are 0.4889 and 0.5798, respec-³⁸⁴ tively. Then, the ten fuzziness levels for testing are generated ³⁸⁵ as level 1 = [0.4889, 0.4980), level 2 = [0.4980, 0.5071), ³⁸⁶ level 3 = [0.5071, 0.5162),..., and level 10 = [0.5707, 0.5798]. ³⁸⁷ Afterwards, we make a statistic for the number of experimen-³⁸⁸ tal trials in each fuzziness level, and plot the histograms as ³⁸⁹ shown in Fig. 1(a) and (c). Finally, we get the average train-³⁹⁰ ing or testing accuracy for each fuzziness level, and plot the ³⁹¹ changing trends as shown in Fig. 1(b) and (d).

One can see from Fig. 1 that the relationship between 392 393 accuracy and fuzziness of ELM does exist for Spam. We ³⁹⁴ further calculate the Pearson correlation coefficient. As a remark, Pearson correlation reflects the statistical relation-395 396 ship between two sets of variables with a coefficient from [-1, 1]. A positive/negative coefficient represents that the 397 398 two sets of variables are positive/negative correlated, and ³⁹⁹ the absolute value represents the correlation degree. We use 400 the median to represent each fuzziness level. Taking the testing result as an example, the correlation coefficient is 401 402 calculated between fuzziness vector [0.4935, 0.5025, 0.5116, 403 0.5207, 0.5298, 0.5389, 0.5480, 0.5571, 0.5662, 0.5753] and 404 accuracy vector [0.8536, 0.8391, 0.8279, 0.8263, 0.8214, 405 0.8194, 0.8177, 0.8111, 0.8065, 0.7524]. Finally, the corre-406 lation coefficients for training and testing are calculated as -0.7145 and -0.8625, respectively. This tells that the accu-407 408 racy and fuzziness have a negative correlation for Spam, 409 i.e., a higher fuzziness will lead to a lower accuracy, and the 410 correlation degree is high.

Although the above example demonstrates that the relation-412 ship between generalization and uncertainty does exist for data 413 set *Spam*, this relationship is difficult to express explicitly for 414 general cases. In the subsequent sections, we will attempt to 415 make this relationship clear by incorporating a new index, 416 i.e., complexity of classification.

417 IV. COMPLEXITY OF CLASSIFICATION PROBLEM

Generally, a classification problem can be described as fol-419 lows. Let S be the universal space we consider, F be a discrete 420 function defined on S. For simplicity, we suppose that func-421 tion F takes values either 0 or 1, where 0 denotes one class 422 and 1 denotes the other class. Given a subset of S, denoted 423 as \mathbb{X} , which is called the training set, the values of F on 424 \mathbb{X} are known, but the values of F on $S - \mathbb{X}$ are unknown. A classification problem is to find a function f such that f can 425 well approximate F both in X and S - X. Usually, F is called 426 an objective function, f is called a classifier acquired based on 427 training set X, the approximation error on X is called training 428 error, and the approximation error on S - X represents the 429 generalization ability of F. 430

The complexity of a classification problem refers to the ⁴³¹ complexity of function F, which implies the difficulties of the ⁴³² process of finding a quality f from X. Unfortunately, there is ⁴³³ no formal definition on the complexity of a discrete function. ⁴³⁴ From references we can find a number of indexes to describe ⁴³⁵ the complexity from different angles. It is noteworthy that the ⁴³⁶ complexity of objective function is independent on the learned ⁴³⁷ classifier f. Since the objective function F is unknown in real ⁴³⁸ applications but is known on the training set X, the indexes ⁴³⁹ in describing the complexity of F can be estimated through ⁴⁴⁰ the training set X and values of F on X. In the following, we ⁴⁴¹ give several indexes to describe the complexity of F, which ⁴⁴² are mainly chosen from [21].

A. Fisher's Discriminant Ratio 444

Fisher's discriminant ratio is an old statistical index for ⁴⁴⁵ describing the difference between two populations. Suppose ⁴⁴⁶ that μ_{1j} , μ_{2j} , σ_{1j} , and σ_{2j} are the means and variances of ⁴⁴⁷ the two populations (classes) with respect to the *j*th attribute, ⁴⁴⁸ *j* = 1, ..., *n*. Then, the Fisher's discriminant ratio for the *j*th ⁴⁴⁹ attributes is defined as ⁴⁵⁰

$$\mathfrak{f}_j = \frac{\left(\mu_{1j} - \mu_{2j}\right)^2}{\sigma_{1j}^2 + \sigma_{2j}^2}.$$
 (10) 451

It is easy to see that Fisher's discriminant ratio with respect 452 to the *j*th attribute describes the distance between two classes 453 regarding this attribute. Intuitively, the longer the distance is, 454 the easier the classification problem is, the lower the complexity will be. Thus, the complexity evaluating index is 456 defined as

$$\mathfrak{Comp}_1 = \frac{1}{\max_j \{\mathfrak{f}_j\}}.$$
(11) 458

459

B. Volume of Overlap Region

A similar measure is the volume of overlap region between 460 two class conditional distributions. It depends on, for each 461 attribute, the maximum and the minimum values of each class. 462



Fig. 2. Intuitive illustration of volume of overlap region.

⁴⁶³ We denote A_j as the *j*th attribute. Then, the overlap region ⁴⁶⁴ normalized by the range of the value spanned by both classes, ⁴⁶⁵ for each attribute A_j , can be represented as

466 \mathfrak{v}_j

$${}^{467} = \frac{\mathrm{MIN}(\max(A_j, c_1), \max(A_j, c_2)) - \mathrm{MAX}(\min(A_j, c_1), \min(A_j, c_2))}{\mathrm{MAX}(\max(A_j, c_1), \max(A_j, c_2)) - \mathrm{MIN}(\min(A_j, c_1), \min(A_j, c_2))}$$

468

$$(12)$$

⁴⁶⁹ where $\max(A_j, c_1)$, $\max(A_j, c_2)$, $\min(A_j, c_1)$, and $\min(A_j, c_2)$ ⁴⁷⁰ denotes the maximum and minimum values of attribute A_j in ⁴⁷¹ the two classes, respectively. Then, the complexity evaluating ⁴⁷² index is defined as the volume of overlap region incorporating ⁴⁷³ all the attributes

474
$$\mathfrak{Comp}_2 = \prod_{j=1}^n \mathfrak{v}_j \tag{13}$$

⁴⁷⁵ where *n* is the number of attributes. An intuitive illustration of ⁴⁷⁶ volume of overlap region for a 2-D feature space is given in ⁴⁷⁷ Fig. 2. It is noted that $\mathfrak{Comp}_2 = 0$ if the value ranges of the two ⁴⁷⁸ classes do not overlap in at least one dimension. Obviously, a ⁴⁷⁹ larger value of \mathfrak{Comp}_2 represents a higher complexity of the ⁴⁸⁰ classification problem.

481 C. Intraclass/Interclass Distance Ratio

This measure first computes the Euclidean distance from each sample to its nearest neighbor within or outside the class. Assume that d_i^{intra} or d_i^{inter} is the distance between sample \mathbf{x}_i and its nearest neighbor within or outside the class, we have

$$\begin{cases} d_i^{\text{intra}} = \min_{j \neq i, y_j = y_i} d(\mathbf{x}_i, \mathbf{x}_j) \\ d_i^{\text{inter}} = \min_{j \neq i, y_j \neq y_i} d(\mathbf{x}_i, \mathbf{x}_j) \end{cases}$$
(14)

⁴⁸⁷ where y_i and y_j represent the class labels of \mathbf{x}_i and \mathbf{x}_j , respec-⁴⁸⁸ tively. Then, it takes the average of all the intraclass distances ⁴⁸⁹ and the average of all the interclass distances, and the ratio of ⁴⁹⁰ both averages is defined as the complexity of the problem

491
$$\mathfrak{Comp}_{3} = \frac{\sum_{i=1}^{N} d_{i}^{\text{intra}}}{\sum_{i=1}^{N} d_{i}^{\text{inter}}}$$
(15)

⁴⁹² where *N* is the number of samples. Similarly, a larger value ⁴⁹³ of \mathfrak{Comp}_3 represents a higher complexity of the classification ⁴⁹⁴ problem.



Fig. 3. Two normal populations.

m

problem can be found from [21].

D. Linear Separability

Linear separability was intensively discussed in the early 496 literature. A simple definition to describe the linear separability for both separable and nonseparable cases is given by 498 Smith [42] 499

in
$$\mathbf{a}^{\mathrm{T}}\mathbf{t}$$
, s.t. $\mathbf{Z}^{\mathrm{T}}\mathbf{w} = \mathbf{t} \ge \mathbf{b}$ (16) 500

where **a** and **b** are arbitrary constant vectors, **w** is the weight ⁵⁰¹ vector, $\mathbf{t} \ge 0$ is the error vector, and **Z** is a matrix in which ⁵⁰² each column **z** is defined based on the input vector **x** and its ⁵⁰³ class label **c** ⁵⁰⁴

$$\mathbf{z} = +\mathbf{x} \quad \text{if } \mathbf{c} = c_1$$

$$\mathbf{z} = -\mathbf{x} \quad \text{if } \mathbf{c} = c_2.$$
(17) 505

The value of the objective function denotes the degree of being 506 separable for two class cases, that is 507

$$\mathfrak{Comp}_4 = \mathbf{a}^{\mathrm{T}} \mathbf{t}. \tag{18} \quad 508$$

It is noted that $Comp_4 = 0$ if the problem is linear separable. 509 Other indexes to describe the complexity of classification 510

In this section, we give an analysis on the relationship 515 between generalization and uncertainty by incorporating the 516 complexity of classification. Since it is difficult for us to give 517 a general analysis for all the complexity indexes, we only 518 adopt the index of Fisher's discriminant ratio in Section IV-A, 519 and give an explanation from the viewpoint of discriminant 520 analysis, which has the principal of maximum probability. 521

Without loss of generality, we consider the 1-D case, which 522 can be easily extended to multiple-dimensional cases. A normal distribution with mean μ and variance σ^2 , denoted by 524 $N(\mu, \sigma^2)$, has a probability density function 525

$$f(x) = \frac{1}{\sqrt{2\pi}\sigma} \exp\left(-\frac{(x-\mu)^2}{2\sigma^2}\right), \ -\infty < x < +\infty.$$
(19) 526

Suppose that there are two normal populations denoted by ${}_{527}$ $N(\mu_1, \sigma_1^2)$ and $N(\mu_2, \sigma_2^2)$ as shown in Fig. 3, and $x(\mu_1 < x < {}_{528}$ $\mu_2)$ is a new sample that needs to be discriminated. ${}_{529}$

For a classification problem, each population represents a $_{530}$ class. From traditional textbook [43] we can view a simple $_{531}$ way to judge sample *x* belonging to which class. $_{532}$

495

Let *C* be the cross-point between two density functions, $_{534}$ i.e., *C* satisfies the following equation:

⁵³⁵
$$\frac{1}{\sqrt{2\pi}\sigma_1} \exp\left(-\frac{(C-\mu_1)^2}{2\sigma_1^2}\right) = \frac{1}{\sqrt{2\pi}\sigma_2} \exp\left(-\frac{(C-\mu_2)^2}{2\sigma_2^2}\right).$$
⁵³⁶ (20)

⁵³⁷ It is easy to check that the cross-point locates in the interval ⁵³⁸ (μ_1, μ_2). The probabilities of sample *x* belonging to the two ⁵³⁹ classes, denoted as (α, β), can be approximately viewed as

540
$$(\alpha, \beta) = \left(\frac{1}{\sqrt{2\pi}\sigma_1} \exp\left(-\frac{(x-\mu_1)^2}{2\sigma_1^2}\right), \frac{1}{\sqrt{2\pi}\sigma_2} \exp\left(-\frac{(x-\mu_2)^2}{2\sigma_2^2}\right)\right)$$
 (21)

⁵⁴² which induces the following discriminant rules based on the ⁵⁴³ principle of maximum probability.

1) IF x < C ($\alpha > \beta$) THEN x belongs to class I.

545 2) IF x > C ($\alpha < \beta$) THEN x belongs to class II.

546 3) IF x = C ($\alpha = \beta$) THEN the class of x is uncertain.

We now relate these discussions about discriminant analysis 547 548 to the theme of this paper, i.e., uncertainty and complexity of 549 a classification problem. According to Section IV-A, the com-550 plexity of a classification problem can be described by means 551 and variances of class distributions. It can be roughly sum-552 marized as: the complexity is going up with either increasing 553 the variances (σ_1^2, σ_2^2) or decreasing the difference between 554 both means $|\mu_1 - \mu_2|$. Moreover, the uncertainty of a classifier is evaluated based on the probability vector (α , β) defined 556 in (21). According to Section III, there are many specific 557 formulas to evaluate the uncertainty (e.g., the fuzziness in 558 Definition 3), but all of them have to satisfy the conditions 559 given in Section III-B, e.g., if $\alpha < \beta$, when $\alpha' < \alpha$ and $\beta' > \beta$, 560 the uncertainty output by vector (α', β') should be smaller than that output by (α, β) . It shows that, to some extent, the differ-561 ⁵⁶² ence between the two probability values denotes the magnitude of uncertainty. The bigger the difference is, the smaller the 563 uncertainty is. Based on these analyses, we have the following 564 565 theorems.

566 Theorem 1: Let

$$_{567} g(\sigma) = \frac{1}{\sqrt{2\pi}\sigma} \left(\exp\left(-\frac{(x-\mu_2)^2}{2\sigma^2}\right) - \exp\left(-\frac{(x-\mu_1)^2}{2\sigma^2}\right) \right)$$

⁵⁶⁸ where $\sigma > 0$, $\mu_1 < \mu_2$, $x \in ([(\mu_1 + \mu_2)/2], \mu_2)$, and μ_1, μ_2 ⁵⁶⁹ are considered as constants. Then, there exists a number $\sigma_1 \in$ ⁵⁷⁰ $(0, \mu_2 - x)$ such that $g(\sigma)$ is monotonically decreasing in the ⁵⁷¹ interval $(\sigma_1, +\infty)$.

Proof: The proof of Theorem 1 is listed in Appendix B.
Theorem 2: Let

574
$$q(\delta) = \frac{1}{\sqrt{2\pi}} \left(\exp\left(-\frac{x - (\mu_2 - \delta)^2}{2}\right) - \exp\left(-\frac{x - (\mu_1 + \delta^*)^2}{2}\right) \right)$$

⁵⁷⁶ where x, μ_1 , and μ_2 are considered as constants, $\mu_1 < \mu_2$, ⁵⁷⁷ $\delta^* = |[(\mu_1 - x)/(\mu_2 - x)]|\delta$, and $\delta > 0$. Then, there exists a number δ_1 such that $q(\delta)$ is monotonically decreasing in the 578 interval $(0, \delta_1)$.

Proof: The proof of Theorem 2 can be derived similarly to 580 the proof of Theorem 1.

Theorem 3: Suppose that the conditional probability outputs of a binary classifier follow two normal distributions $N(\mu_1, \sigma^2)$ and $N(\mu_2, \sigma^2)$, respectively, where $\mu_1 < \mu_2$. Let 584

$$\alpha = -\frac{1}{\sqrt{2\pi}\sigma} \exp\left(-\frac{(x-\mu_1)^2}{2\sigma^2}\right)\beta$$

$$= 1 \qquad (x-\mu_2)^2)$$
585

$$= \frac{1}{\sqrt{2\pi\sigma}} \exp\left(-\frac{(\chi - \mu_2)}{2\sigma^2}\right)$$
 580

and

$$E(\alpha, \beta) = -\frac{1}{2}(\alpha \log \alpha + (1 - \alpha) \log(1 - \alpha))$$
580

$$+ \beta \log \beta + (1 - \beta) \log(1 - \beta)).$$
 589

Assume $\beta = K\alpha$ where $K \in (1, 1 + \epsilon)$, then $E(\alpha, \beta) = {}^{590}$ E(K) is monotonically decreasing with respect to K if 591 $K\alpha > (1/2)$.

Proof: The proof of Theorem 3 is listed in Appendix C. Noting that $g(\sigma)$ in Theorem 1 or $q(\delta)$ in Theorem 2 denotes the difference between two probability density values, which can be represented as $\beta - \alpha$ in Theorem 3. Theorem 3 directly connects this difference together with the uncertainty of the classifier's outputs given in Definition 2.

Theorem 3 shows that the uncertainty of the classifier's out- 599 puts is decreasing with the increase of the difference between 600 two density values, i.e., $\beta - \alpha$, where α and β can be con- 601 sidered as the probabilities of a sample being classified as 602 classes I and II, respectively. As a result, the conclusions in 603 Theorems 1 and 2 show that the uncertainty of a classifier's 604 outputs is becoming bigger with the increase of the complex- 605 ity of the classification problem, which is represented through 606 inflating the variance in Theorem 1 and through shrinking 607 the difference between two means in Theorem 2, respectively. 608 Since in a classification problem, the complexity is inherent 609 while the uncertainty is generated by the output of a well- 610 trained classifier which has its training and testing accuracy, 611 it is reasonable to believe that some relationships exist among 612 the accuracy, uncertainty, and complexity. 613

It is noteworthy that Theorems 1–3 cannot exactly explain ⁶¹⁴ the relationships among the three indexes, i.e., accuracy, ⁶¹⁵ uncertainty, and complexity. However, to a great extent, ⁶¹⁶ they provide solid supports to the existence of the relationships. They confirm such a fact that the classifier's uncertainty will be inevitably high if the classification problem ⁶¹⁹ is complex, no matter what classifier design algorithm is ⁶²⁰ used. This statement further implies that a high-performance ⁶²¹ classifier will have high uncertainty when the problem is ⁶²² complex. ⁶²³

VI. EMPIRICAL STUDIES 624

In this section, we will conduct some empirical studies to 625 further analyze the relationships discussed in Section V. It 626 is noteworthy the discussions in Section V were made based 627

655

679

680

681

TABLE I Selected Data Sets for Experiments

No	Data Set	# Samples	# Features	# Classes
1	Libras	369	90	15
2	Breast	699	9	2
3	SPECTF	267	44	2
4	Cancer	683	9	2
5	Chart	600	60	6
6	Cotton	356	21	6
7	CT	221	36	2
8	Dermatology	366	34	6
9	Ecoli	336	7	8
10	German	1,000	24	2
11	Glass	214	9	6
12	Haberman	306	3	2
13	Heart	270	13	2
14	Vowel	990	10	11
15	Ionosphere	351	34	2
16	Australian	690	14	2
17	Pima	768	8	2
18	Plrx	182	12	2
19	Sonar	208	60	2
20	Soybean	683	35	19
21	Bupa	345	6	2
22	Transfusion	748	4	2
23	Segment	2,310	19	7
24	Wdbc	569	30	2
25	Wpbc	198	33	2
26	Yeast	1,484	8	10
27	Zoo	101	16	7
28	Spam	4,601	57	2
29	Satellite	6,435	36	6
30	OptDigits	5,620	64	10
31	Pen	10,992	16	10

⁶²⁸ on \mathfrak{Comp}_1 , i.e., Fisher's discriminant ratio. Thus, in this sec-⁶²⁹ tion, we will also adopt \mathfrak{Comp}_1 to evaluate the complexity of ⁶³⁰ classification problems.

631 A. Selected Data Sets

The data sets used in the experiments are selected from UCI machine learning repository. The detailed information regarding these data sets are summarized in Table I. Since the complexity indexes listed in Section IV are defined for binary classification problems, we transfer each multiclass data set into binary by randomly selecting 50% classes as positive and the rest 50% classes as negative.

639 B. Experimental Design

The flowchart for training the classifier and evaluating the problem complexity is listed in Algorithm 1.

It is noteworthy that the training algorithm adopted in this section is ELM. Due to the random mechanism for weight assignment, it is easy to repeat the experiment for many times. We conduct 100 experimental trials for each data set. In each trial, 70% data are randomly selected for training, and the remaining 30% data are used for testing. Each trial will protable vide a different result, and we make statistics for fuzziness, accuracy, and complexity based on the 100 results.

The number of hidden nodes in ELM is set as 20, and sigmoid activation function is utilized. The simulations are carried out under MATLAB R2011b, which are executed on a computer with an Intel Core i7-5500U CPU@2.40 GHz, 8GB memory, and 64-bit Windows 8 system.

Algorithm	1:	Train	ELM	Classifier	and	Compute
Evaluating I	ndey	xes				

Input:
Training set $\mathbb{X} = \{(\mathbf{x}_i, \mathbf{t}_i)\}_{i=1}^N \subset \mathcal{R}^n \times \{0, 1\}^c;$
Activation function $f(\mathbf{x})$;
Number of hidden nodes \tilde{N} .
Output:
Fuzziness and generalization of the trained classifier;
Complexity of the classification problem.
Data processing: randomly divide the data set into two
parts for training and testing according to a separation

- ratio.2 *Classifier training*: train a ELM classifier based on the algorithm given in section II-A.
- 3 *Testing*: test the classifier on the testing set, compute the fuzziness (Definition 3) and generalization (testing accuracy) of the classifier.
- 4 *Complexity evaluation*: compute the complexity of the classification problem, i.e., Eq. (11).

C. Experimental Analysis

1

Similar to Section III-C, we make some statistical analyses 656 on the testing results. For each data set, ten fuzziness lev- 657 els are generated by equally dividing the interval between 658 the maximum and minimum fuzziness values. We use the 659 median to represent each fuzziness level. Then, the number 660 of experimental trials for each fuzziness level is counted, 661 and the average testing accuracy for each fuzziness level is 662 calculated. Fig. 4 demonstrates the changing trend of the test- 663 ing accuracy along with the level of fuzziness. It depicts 664 the dependency relation between testing accuracy and test- 665 ing fuzziness for the classification problems. Due to space 666 limit, we only plot the results for 12 data sets out of 31. 667 Furthermore, we calculate the Pearson correlation coefficient 668 between fuzziness vector and accuracy vector for each data 669 set. It is noteworthy there are ten fuzziness levels for each 670 data set. However, from Fig. 4, we can see that the high- 671 est fuzziness level (i.e., level ten) usually cause a sharp 672 change of the testing accuracy, which may interfere the sta- 673 tistical analysis for the overall results. Thus, we only use the 674 previous nine fuzziness values and their corresponding accu- 675 racy. The correlation coefficients r are listed in Table II. We 676 artificially set up some thresholds to justify the degree of 677 correlation. 678

1) If $0 \le |r| < 0.4$, then the correlation is low.

2) If $0.4 \le |r| < 0.7$, then the correlation is medium.

3) If $0.7 \le |r| \le 1$, then the correlation is strong.

It is observed from Table II that the generalization and fuzziness have a strong or medium correlation regarding most data sets. 682

The complexities of the problems are shown in Fig. 5, 685 which are sorted according to the order numbers (i.e., 1–31) 686 in Table I. In Fig. 5, we artificially set up a threshold such 687 that the complexity higher than the threshold is called high 6688 otherwise is called low. In this case, one can view an implicit 689 relation among the complexity, generalization, and fuzziness. 690



Fig. 4. Relationship between fuzziness and generalization of ELM classifier on different data sets. (a) Australian. (b) Chart. (c) Dermatology. (d) Segment. (e) Libras. (f) OptDigits. (g) Pen. (h) Plrx. (i) Sonar. (j) Spam. (k) SPECTF. (l) Yeast.

Data	Pearson Correlation	Data	Pearson Correlation
Set	Coefficient	Set	Coefficient
1	-0.6434	17	-0.0520†
2	-0.3522†	18	0.9743★
3	0.7838★	19	-0.6896
4	-0.7835★	20	-0.9348★
5	-0.7718★	21	0.4132
6	0.3421†	22	0.4962
7	0.3744†	23	0.8728★
8	-0.9277★	24	-0.2933†
9	-0.0579†	25	-0.5559√
10	-0.1474†	26	0.6297
11	-0.5452	27	0.3470†
12	0.5803	28	-0.9496★
13	0.1768†	29	0.1455†
14	-0.9362★	30	-0.9903★
15	0.5782	31	-0.9895★
16	0.7420+		

TABLE II PEARSON CORRELATION COEFFICIENT BETWEEN OUTPUT FUZZINESS AND TESTING ACCURACY

Note: For each data set, \star represents strong correlation, $\sqrt{}$ represents medium correlation, and \dagger represents low correlation.

⁶⁹¹ The generalization of a classifier trained by ELM goes up ⁶⁹² with the increase of fuzziness if the complexity of the clas-⁶⁹³ sification problem is relatively high, while the generalization ⁶⁹⁴ of a classifier trained by ELM goes down with the increase ⁶⁹⁵ of fuzziness if the complexity of the classification problem is



Fig. 5. Complexity of the classification problems.

relatively low. For instance, it can be seen from Fig. 5 that ⁶⁹⁶ the complexity values of *Segment* (data set 23) and *Plrx* (data ⁶⁹⁷ set 18) are high, in this case, the generalizations of these two ⁶⁹⁸ data sets are becoming better with the increase of fuzziness ⁶⁹⁹ as shown in Fig. 4(d) and (h). However, the complexity val-⁷⁰⁰ ues of *OptDigits* (data set 30) and *Spam* (data set 28) are ⁷⁰¹ low, in this case, the generalizations of these two data sets are ⁷⁰² becoming worse with the increase of fuzziness as shown in ⁷⁰³ Fig. 4(f) and (j). ⁷⁰⁴

By learning the complexity of classification problems from $_{705}$ Fig. 5, we grasp some factors that are resulted from the $_{706}$



Fig. 6. Relationship between fuzziness and generalization of SVM classifier on different data sets. (a) Australian. (b) Chart. (c) Dermatology. (d) Segment. (e) Libras. (f) OptDigits. (g) Pen. (h) Plrx. (i) Sonar. (j) Spam. (k) SPECTF. (l) Yeast.

707 complexity of decision boundaries. It is obvious that there are708 some relations between them.

As we know, the complexity of a classification problem room to can be intuitively regarded as the degree of difficulty for the room problem. More specifically, it is the complexity of geometrical room that divides the sample space. In classification problem, it is room that divides the sample space. In classification problem, it is room that divides the sample space. In classification problem, it is room that divides the sample space. In classification problem, it desired to find a classifier f by training the data set locating room to the boundary function F = 0. The ability of function room to the function F on unseen data is the generalizaroom to the function F on unseen data is the generalizaroom to the function f in dividing unseen samples.

⁷¹⁹ When it is easy to distinguish the classes by the boundary of ⁷²⁰ function *F*, it will also be easy to divide the unseen samples by ⁷²¹ *f*, since the structure of training data is supposed to be similar ⁷²² to the structure of unseen data and *f* is an estimator of *F*. ⁷²³ It implies that the boundary will be simple and the fuzziness ⁷²⁴ of the boundary is low. In this situation, it is reasonable to ⁷²⁵ believe that, with the decrease of classifier's fuzziness, the ⁷²⁶ generalization will be improved.

⁷²⁷ When it is difficult to distinguish the classes by the bound-⁷²⁸ ary of function F, the classifier function f is also difficult to ⁷²⁹ divide the unseen samples. It corresponds to a case of high ⁷³⁰ complexity and complex boundary. It is inherent to output ⁷³¹ high fuzziness for boundary samples for any classifier, and therefore, we reasonably believe in this situation that, with ⁷³² the increase of classifier's fuzziness, the generalization may ⁷³³ be getting better. ⁷³⁴

D. Analysis With SVM Classifiers

We further realize the above studies with SVM classifiers. ⁷⁹⁶ We adopt the "LibSVM" toolbox, the penalty term *C* is fixed ⁷³⁷ as 100, and RBF kernel $\mathcal{K}(\mathbf{x}, \mathbf{x}_i) = \exp(-[||\mathbf{x} - \mathbf{x}_i||^2/2\sigma^2])$ ⁷³⁸ with $\sigma = 1$ is adopted. The decision values of SVM are ⁷³⁹ transformed into uncertain outputs by logistic function. The ⁷⁴⁰ dependency relation between generalization and fuzziness ⁷⁴¹ regarding the 12 data sets in Fig. 4 are demonstrated in Fig. 6. ⁷⁴² It can be observed that the results are basically consistent with ⁷⁴³ those in Section VI-C, but the changing trends are not as clear ⁷⁴⁴ as those of ELM. As a result, ELM might be more suitable to ⁷⁴⁵ conduct this paper, since it has a higher degree of uncertainty ⁷⁴⁶ due to the random mechanism for input weights assignment. ⁷⁴⁷

VII. CONCLUSION 748

735

This paper finds an empirical relationship among the complexity of a classification problem, the uncertainty of classifier's outputs, and the prediction accuracy of the classifier. By experimental validation and theoretical explanation through a simple model of discriminant analysis, it is found that with the ⁷⁵⁴ increase of the uncertainty of the classifier's outputs, empiri-⁷⁵⁵ cally the accuracy is upgrading for high-complexity problem ⁷⁵⁶ but downgrading for low-complexity problem. Based on these ⁷⁵⁷ findings, in order to choose a better classification rule for a ⁷⁵⁸ practical problem, one can tune the model parameters such that ⁷⁵⁹ the uncertainty becomes larger for problems with higher com-⁷⁶⁰ plexity, or smaller for problems with lower complexity under ⁷⁶¹ the condition that an acceptable training accuracy is kept.

762 APPENDIX A 763 FEATURES OF ELMS

⁷⁶⁴ In the following, we briefly review the major advantages ⁷⁶⁵ of ELMs.

- The first advantage of ELMs is the fast training speed.
 Since the training of ELMs does not include iterative tuning, it statistically shows that ELM is thousands of times faster than BP given a predefined threshold for training accuracy.
- 2) Another feature of ELMs is the acceptable generalization ability. In comparison with other popular classification or regression algorithms, such as DTs, SVMs,
 logistic regressions, etc., the generalization of ELMs
 may not be the best in general. But so far, one cannot
 find a significant difference among the generalizations
 of these algorithms.
- The training procedure of ELMs can process online
 sequential data conveniently, which demonstrates strong
 potentials for big data analytic. It is shown that ELMs
 can effectively handle both numerical and nominal
 attributes for both classification and regression problems.

4) Mathematically it is proven that ELMs have the universal approximation ability if the activation function is differentiable. That is, ELMs can uniformly approximate any continuous function defined in an interval when the number of hidden nodes goes to infinity. This conclusion establishes the foundation of applying ELMs to various classification and regression problems.

It is worthy noting that any learning algorithm cannot be
 consistently better than others. In the following, we list several
 disadvantages of ELMs.

 As aforementioned, the weights between input and hidden layers in ELMs are randomly selected from an interval. ELMs are sensitive to this interval, and the change of the interval will produce quite different classifiers, which seriously decreases the stability.

The number of hidden layer nodes is critical for building
an ELM. A large number will lead to the generalization
decreasing but a small number can result in the training
error increasing. So far, how to select the number of
hidden layer nodes is still a challenging issue.

803Appendix B804Proof of Theorem 1

⁸⁰⁵ The original problem can be represented as

$$g(\sigma) = \frac{1}{\sqrt{2\pi}\sigma} \left(\exp\left(-\frac{(x-b)^2}{2\sigma^2}\right) - \exp\left(-\frac{(x-a)^2}{2\sigma^2}\right) \right)$$

prove that there exits σ_1 such that $g(\sigma)$ is monotonically ⁸⁰⁷ increasing when $\sigma < \sigma_1$ and $g(\sigma)$ is monotonically decreasing ⁸⁰⁸ when $\sigma > \sigma_1$.

The constant term $\sqrt{2\pi}$ can be neglected. Let $(x - a) = {}_{810}$ $k \times (b - x)$ and $\sigma = t \times (b - x)$, the original problem can be ${}_{811}$ simplified as

$$g(t) = \frac{1}{t} \left(\exp(-\frac{1}{2t^2}) - \exp(-\frac{k^2}{2t^2}) \right), \quad k > 1 \text{ and } t > 0$$
 813

prove that there exits t_1 such that g(t) is monotonically ⁸¹⁴ increasing when $t < t_1$ and g(t) is monotonically decreasing ⁸¹⁵ when $t > t_1$.

We get the first-order derivation of g(t), that is

$$g'(t) = \frac{1}{t^4} \left[\left(1 - t^2 \right) \exp\left(-\frac{1}{2t^2} \right) - \left(k^2 - t^2 \right) \exp\left(-\frac{k^2}{2t^2} \right) \right].$$
 B1B

Having this derivation, it can be derived as follows.

- 1) When t > k, $t^2 1 > t^2 k^2 > 0$ and $\exp(-[1/2t^2]) > \frac{1}{2} \exp(-[k^2/2t^2])$, thus $(t^2 1)\exp(-[1/2t^2]) > (t^2 \frac{1}{2})$ $k^2)\exp(-[k^2/2t^2])$, thus we have g'(t) < 0.
- 2) When $k \ge t > 1$, $(1 t^2) \exp(-[1/2t^2]) < 0$, thus ⁸²³ $(k^2 - t^2) \exp(-[k^2/2t^2]) > 0$, thus we have g'(t) < 0. ⁸²⁴ 3) When t = 1, we have $g'(t) = [1/t^4][-(k^2 - 825)t^2) \exp(-[k^2/2t^2])] < 0$.

So far, we have proved that g'(t) < 0 when $t \ge 1$, which B27 means that g(t) is monotonically decreasing when $t \ge 1$. B28

When 1 > t > 0 and $t \to 0$, we have $[(1 - t^2)/(k^2 - t^2)] \to 629$ $(1/k^2)$ and $\exp([(1 - k^2)/2t^2]) \to 0$ (noting that $t \le 1 < k$). 630 There exists $t^* \in (0, 1)$ such that $([(1 - t^{*2})/(k^2 - t^{*2})] > 831$ $\exp([(1 - k^2)/2t^{*2}]) = [\exp(1/2t^{*2})/\exp([k^2/2t^{*2}])]$, thus 632 $([(1 - t^{*2})\exp(-1/2t^{*2})]/[(k^2 - t^{*2})\exp(-k^2/2t^{*2})]) > 1$, 633 thus $(1 - t^{*2})\exp(-1/2t^{*2}) > (k^2 - t^{*2})\exp(-k^2/2t^{*2})$, thus 634 $g'(t^*) > 0$. 635

According to Zero theorem, there exits $t_1 \in (0, 1)$ such that ⁸³⁶ $g'(t_1) = 0$. Since g'(t) is continuous and differentiable, if all ⁸³⁷ the stagnation points are maximum points, then there is only ⁸³⁸ one stagnation point, otherwise minimum point exists. ⁸³⁹

We further get the second-order derivation of g(t), that is 840

$$g''(t) = \frac{1}{t^7} \left\{ \left[2t^2 \left(t^2 - 1 \right) - 2t^2 + (1 - t^2) \right] \exp\left(-\frac{1}{2t^2} \right) \right\}$$

$$-\left[2t^{2}\left(t^{2}-k^{2}\right)-2t^{2}k^{2}+k^{2}\left(k^{2}-t^{2}\right)\right]\exp\left(-\frac{k}{2t^{2}}\right)\right\}.$$
 842

Put the stagnation point t_1 into g''(t), since $(1 - t_1^2)$ ⁸⁴³ exp $(-1/2t_1^2) - (k^2 - t_1^2) \exp(-k^2/2t_1^2) = 0$, we have ⁸⁴⁴

$$g''(t_1) = \frac{1}{t_1^7} \left\{ -2t_1^2 \left[\exp\left(-\frac{1}{2t_1^2}\right) - k^2 \exp\left(-\frac{k^2}{2t_1^2}\right) \right] \right\}^{845} + \left(1 - t_1^2\right) \exp\left(-\frac{1}{2t_1^2}\right) - k^2 \left(k^2 - t_1^2\right) \exp\left(-\frac{k^2}{2t_1^2}\right) \right\}^{845}$$

+
$$\left(1 - t_{1}^{2}\right) \exp\left(-\frac{1}{2t_{1}^{2}}\right) - k^{2}\left(k^{2} - t_{1}^{2}\right) \exp\left(-\frac{k^{2}}{2t_{1}^{2}}\right)$$
. 846

Based on

$$\left(1 - t_1^2\right) \exp\left(-\frac{1}{2t_1^2}\right) - \left(k^2 - t_1^2\right) \exp\left(-\frac{k^2}{2t_1^2}\right) = 0 \qquad \text{848}$$

$$k > 1 \text{ and } 1 > t_1 > 0 \qquad \text{849}$$

817

819

850 we have

$$\sup \left(-\frac{1}{2t_1^2} \right) - k^2 \exp \left(-\frac{k^2}{2t_1^2} \right)$$
$$= t_1^2 \left[\exp \left(-\frac{1}{2t_1^2} \right) - \exp \left(-\frac{k^2}{2t_1^2} \right) \right]$$

854 and

855
$$\left(1 - t_1^2\right) \exp\left(-\frac{1}{2t_1^2}\right) - k^2 \left(k^2 - t_1^2\right) \exp\left(-\frac{k^2}{2t_1^2}\right)$$

856
$$< \left(1 - t_1^2\right) \exp\left(-\frac{1}{2t_1^2}\right) - \left(k^2 - t_1^2\right) \exp\left(-\frac{k^2}{2t_1^2}\right)$$

857
$$= 0.$$

⁸⁵⁸ Thus, $g''(t_1) < 0$, t_1 is the maximum point, which means ⁸⁵⁹ that g(t) is monotonically increasing when $t < t_1$ and g(t) is ⁸⁶⁰ monotonically decreasing when $t_1 < t < 1$.

To this end, we have proved that g(t) is monotonically increasing when $t < t_1$ and g(t) is monotonically decreasing when $t > t_1$.

864 APPENDIX C

1

Proof of Theorem 3

Substituting β with $K\alpha$ in E(K), we have

$$E(K) = -\frac{1}{2} \left(\alpha \log \alpha + (1 - \alpha) \log(1 - \alpha) + K\alpha \log(K\alpha) + (1 - K\alpha) \log(1 - K\alpha) \right)$$

Taking derivative of E(K) with respect to K, we obtain

$$\frac{dE(K)}{d(K)} = -\frac{1}{2}(\alpha \log(K\alpha) - \alpha \log(1 - K\alpha))$$

871

874

870

865

$$=-\frac{1}{2}\log\frac{1-k}{1-k}$$

⁸⁷² It is easy to view that [dE(K)/d(K)] < 0 if $K\alpha > (1/2)$, ⁸⁷³ which completes the proof.

REFERENCES

- [1] W. W. Y. Ng, A. P. F. Chan, D. S. Yeung, and E. C. C. Tsang,
 "Quantitative study on the generalization error of multiple classifier systems," in *Proc. IEEE Int. Conf. Syst. Man Cybern.*, 2005,
 pp. 889–894.
- [2] C. Bishop, *Pattern Recognition and Machine Learning*. New York, NY,
 USA: Springer-Verlag, 2006.
- [3] X. D. Wu *et al.*, "Top 10 algorithms in data mining," *Knowl. Inf. Syst.*, vol. 14, no. 1, pp. 1–37, 2008.
- [4] Z. Yan and C. Xu, "Studies on classification models using decision boundaries," in *Proc. 8th IEEE Int. Conf. Cogn. Informat.*, Hong Kong, 2009, pp. 287–296.
- [5] R. O. Duda, P. E. Hart, and D. G. Stork, *Pattern Classification*.
 New York, NY, USA: Wiley, 2012.
- [6] V. N. Vapnik, *The Nature of Statistical Learning Theory*. New York,
 NY, USA: Springer-Verlag, 2000.
- [7] R. Wang, S. Kwong, and D. Chen, "Inconsistency-based active learning for support vector machines," *Pattern Recognit.*, vol. 45, no. 10, pp. 3751–3767, 2012.
- [8] R. Wang and S. Kwong, "Active learning with multi-criteria decision making systems," *Pattern Recognit.*, vol. 47, no. 9, pp. 3106–3119, 2014.
- [9] R. Wang, D. Chen, and S. Kwong, "Fuzzy-rough-set-based active learning," *IEEE Trans. Fuzzy Syst.*, vol. 22, no. 6, pp. 1699–1704, Dec. 2014.

- [10] R. Wang, C.-Y. Chow, and S. Kwong, "Ambiguity-based multiclass as active learning," *IEEE Trans. Fuzzy Syst.*, vol. 24, no. 1, pp. 242–248, 899 Feb. 2016.
- J. R. Quinlan, "Induction of decision trees," *Mach. Learn.*, vol. 1, no. 1, 901 pp. 81–106, 1986.
- [12] R. Wang, S. Kwong, X.-Z. Wang, and Q. Jiang, "Segment based decision 903 tree induction with continuous valued attributes," *IEEE Trans. Cybern.*, 904 vol. 45, no. 7, pp. 1262–1275, Jul. 2015. 905
- M. T. Hagan and M. B. Menhaj, "Training feedforward networks with 906 the Marquardt algorithm," *IEEE Trans. Neural Netw.*, vol. 5, no. 6, 907 pp. 989–993, Nov. 1994.
- [14] G.-B. Huang, L. Chen, and C.-K. Siew, "Universal approximation using 909 incremental constructive feedforward networks with random hidden 910 nodes," *IEEE Trans. Neural Netw.*, vol. 17, no. 4, pp. 879–892, Jul. 2006. 911
- [15] K.-I. Funahashi, "On the approximate realization of continuous mappings by neural networks," *Neural Netw.*, vol. 2, no. 3, pp. 183–192, 913 1989.
- [16] G. Cybenko, "Approximation by superpositions of a sigmoidal function," 915 Math. Control Signals Syst., vol. 2, no. 4, pp. 303–314, 1989.
- [17] C.-F. Lin and S.-D. Wang, "Fuzzy support vector machines," *IEEE* 917 *Trans. Neural Netw.*, vol. 13, no. 2, pp. 464–471, Mar. 2002.
- [18] C. Z. Janikow, "Fuzzy decision trees: Issues and methods," *IEEE Trans.* 919 Syst., Man, Cybern. B, Cybern., vol. 28, no. 1, pp. 1–14, Feb. 1998. 920
- [19] Z. Deng, K.-S. Choi, Y. Jiang, and S. Wang, "Generalized hiddenmapping ridge regression, knowledge-leveraged inductive transfer learning for neural networks, fuzzy systems and kernel methods," *IEEE Trans.* 923 *Cybern.*, vol. 44, no. 12, pp. 2585–2599, Dec. 2014. 924
- [20] X.-Z. Wang *et al.*, "A study on relationship between generalization abilities and fuzziness of base classifiers in ensemble learning," *IEEE Trans.* 926 *Fuzzy Syst.*, vol. 23, no. 5, pp. 1638–1654, Oct. 2015.
- [21] T. K. Ho and M. Basu, "Complexity measures of supervised classification problems," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 24, no. 3, 929 pp. 289–300, Mar. 2002. 930
- [22] G.-B. Huang, Q.-Y. Zhu, and C.-K. Siew, "Extreme learning 931 machine: Theory and applications," *Neurocomputing*, vol. 70, nos. 1–3, 932 pp. 489–501, 2006.
- [23] E. Soria-Olivas *et al.*, "BELM: Bayesian extreme learning machine," 934 *IEEE Trans. Neural Netw.*, vol. 22, no. 3, pp. 505–509, Mar. 2011.
- [24] G.-B. Huang, H. M. Zhou, X. J. Ding, and R. Zhang, "Extreme learning 936 machine for regression and multiclass classification," *IEEE Trans. Syst.*, 937 *Man, Cybern. B, Cybern.*, vol. 42, no. 2, pp. 513–529, Apr. 2012.
- [25] A. A. Mohammed, R. Minhas, Q. M. J. Wu, and M. A. Sid-939 Ahmed, "Human face recognition based on multidimensional PCA 940 and extreme learning machine," *Pattern Recognit.*, vol. 44, nos. 10–11, 941 pp. 2588–2597, 2011. 942
- [26] K. A. Toh, "Deterministic neural classification," *Neural Comput.*, 943 vol. 20, no. 6, pp. 1565–1595, 2008.
- [27] X. Liu, S. Lin, J. Fang, and Z. Xu, "Is extreme learning machine feasible? A theoretical assessment (part I)," *IEEE Trans. Neural Netw. Learn.* 946 *Syst.*, vol. 26, no. 1, pp. 7–20, Jan. 2015.
- [28] S. Lin, X. Liu, J. Fang, and Z. Xu, "Is extreme learning machine feasible? A theoretical assessment (part II)," *IEEE Trans. Neural Netw.* 949 *Learn. Syst.*, vol. 26, no. 1, pp. 21–34, Jan. 2015.
- [29] J. Cao, K. Zhang, M. Luo, C. Yin, and X. Lai, "Extreme learning 951 machine and adaptive sparse representation for image classification," 952 *Neural Netw.*, vol. 81, no. C, pp. 91–102, Sep. 2016. 953
- [30] J. Cao, J. Hao, X. Lai, C.-M. Vong, and M. Luo, "Ensemble extreme 954 learning machine and sparse representation classification algorithm," 955 *J. Franklin Inst.*, vol. 353, no. 17, pp. 4526–4541, 2016.
- B. Igelnik and Y.-H. Pao, "Stochastic choice of basis functions in adaptive function approximation and the functional-link net," *IEEE Trans.* 958 *Neural Netw.*, vol. 6, no. 6, pp. 1320–1329, Nov. 1995.
- [32] W. Deng, Q. Zheng, and L. Chen, "Regularized extreme learning 960 machine," in *Proc. IEEE Symp. Comput. Intell. Data Min.*, Nashville, 961 TN, USA, 2009, pp. 389–395.
- [33] H.-J. Rong, Y.-S. Ong, A.-H. Tan, and Z. Zhu, "A fast pruned-extreme 963 learning machine for classification problem," *Neurocomputing*, vol. 72, 964 nos. 1–3, pp. 359–366, 2008.
- [34] G. Feng, G.-B. Huang, Q. Lin, and R. Gay, "Error minimized extreme 966 learning machine with growth of hidden nodes and incremental learning," *IEEE Trans. Neural Netw.*, vol. 20, no. 8, pp. 1352–1357, 968 Aug. 2009. 969
- [35] N.-Y. Liang, G.-B. Huang, P. Saratchandran, and N. Sundararajan, 970 "A fast and accurate online sequential learning algorithm for feedforward 971 networks," *IEEE Trans. Neural Netw.*, vol. 17, no. 6, pp. 1411–1423, 972 Nov. 2006. 973

- 974 [36] G.-B. Huang, Q.-Y. Zhu, and C.-K. Siew, "Extreme learning machine:
 975 A new learning scheme of feedforward neural networks," in *Proc. IEEE*
- Int. Joint Conf. Neural Netw., Budapest, Hungary, 2004, pp. 985–990.
 [37] L. A. Zadeh, "Probability measures of fuzzy events," J. Math. Anal.
- Appl., vol. 23, no. 2, pp. 421–427, 1968.
 [38] A. De Luca and S. Termini, "A definition of a nonprobabilistic entropy in
- the setting of fuzzy sets theory," *Inf. Control*, vol. 20, no. 4, pp. 301–312, 1972.
- ⁹⁸² [39] G. J. Klir, "Where do we stand on measures of uncertainty, ambiguity,
 ⁹⁸³ fuzziness, and the like?" *Fuzzy Set Syst.*, vol. 24, no. 2, pp. 141–160,
 ⁹⁸⁴ 1987.
- 985 [40] G. J. Klir and T. A. Folger, *Fuzzy Sets, Uncertainty and Information*.
 986 Englewood Cliffs, NJ, USA: Prentice-Hall, 1998.
- 987 [41] D. Sánchez and E. Trillas, "Measures of fuzziness under different uses of fuzzy sets," in *Advances in Computational Intelligence*989 (Communications in Computer and Information Science), vol. 298.
 990 Heidelberg, Germany: Springer, 2012, pp. 25–34.
- F. W. Smith, "Pattern classifier design by linear programming," *IEEE Trans. Comput.*, vol. C-17, no. 4, pp. 367–372, Apr. 1968.
- P33 [43] R. O. Duda, P. E. Hart, and D. G. Stork, *Pattern Classification*.
 New York, NY, USA: Wiley, 2001.
- ⁹⁹⁵ [44] S. Ding, N. Zhang, and J. Zhang, "Unsupervised extreme learning
 ⁹⁹⁶ machine with representational features," *Int. J. Mach. Learn. Cybern.*,
 ⁹⁹⁷ vol. 8, no. 2, pp. 587–595, Apr. 2017s.
- 998 [45] S. Balasundaram and D. Gupta, "On optimization based extreme learning machine in primal for regression and classification by functional iterative method," *Int. J. Mach. Learn. Cybern.*, vol. 7, no. 5, pp. 707–728, Oct. 2016.
- P. Liu, Y. Huang, L. Meng, and S. Gong, "Two-stage extreme learning machine for high-dimensional data," *Int. J. Mach. Learn. Cybern.*, vol.
 7, no. 5, pp. 765–772, Oct. 2016.
- I. Zhang, S. Ding, and N. Zhang, "Incremental extreme learning machine based on deep feature embedded," *Int. J. Mach. Learn. Cybern.*, vol. 7,
- no. 1, pp. 111–120, Feb. 2016.
 A. Fu, C. Dong, and L. Wang, "An experimental study on stability and generalization of extreme learning machines," *Int. J. Mach. Learn.*
- 1010 *Cybern.*, vol. 6, no. 1, pp. 129–135, Feb. 2015.



Xi-Zhao Wang (M'03–SM'04–F'12) received the Doctoral degree in computer science from the Harbin Institute of Technology, Harbin, China, in 1998.

From 2001 to 2014, he was a Full Professor and the Dean of the College of Mathematics and Computer Science, Hebei University, Hebei, China. From 1998 to 2001, he was a Research Fellow with the Department of Computing, Hong Kong Polytechnic University, Hong Kong. Since 2014, he has been a Full Professor with the College of Computer Science and Software Engineering,

1022 Shenzhen University, Shenzhen, China. His current research interests include
1023 supervised and unsupervised learning, active learning, reinforcement learning,
1024 manifold learning, transfer learning, unstructured learning, uncertainty, fuzzy
1025 sets and systems, fuzzy measures and integrals, rough set, and learning from
1026 big data.

Dr. Wang was a recipient of many awards from the IEEE SMC Society. Dr. Wang was a recipient of many awards from the IEEE International Conference on Systems, Man, and Cybernetics (SMC) in 2005, 2007–2009, and 2012–2014, the Chair of the Technical Committee on Computational Intelligence of the IEEE SMC, and a Distinguished Lecturer of the IEEE SMC. He was the Program Co-Chair of the IEEE SMC 2009 and 2010. He is the Editor-in-Chief of the *International Journal of* Machine Learning and Cybernetics. He is also an Associate Editor of the IEEE TRANSACTIONS ON SYSTEMS, MAN, AND CYBERNETICS—PART B: Dr. CyBERNETICS, *Information Sciences Journal*, and the *International Journal for Pattern Recognition and Artificial Intelligence*.



Chen Xu received the B.Sc. and M.Sc. degrees 1052 from Xidian University, Xi'an, China, in 1986 and 1053 1989, respectively, and the Ph.D. degree from Xi'an 1054 Jiaotong University, Xi'an, in 1992. 1055

He joined Shenzhen University, Shenzhen, China, 1056 in 1992, where he is currently a Professor. From 1057 1999 to 2000, he was a Research Fellow with 1058 Kansai University, Suita, Japan, and the University 1059 of Hawaii, Honolulu, HI, USA, from 2002 to 2003. 1060 His current research interests include image process-1061 ing, intelligent computing, and wavelet analysis. 1062

Ran Wang (S'09–M'14) received the B.Eng. 1038 degree in computer science from the College 1039 of Information Science and Technology, Beijing 1040 Forestry University, Beijing, China, in 2009, and 1041 the Ph.D. degree from the Department of Computer 1042 Science, City University of Hong Kong, Hong Kong, 1043 in 2014. 1044

From 2014 to 2016, she was a Post-Doctoral 1045 Researcher with the Department of Computer 1046 Science, City University of Hong Kong. She is cur- 1047 rently an Assistant Professor with the College of 1048

Mathematics and Statistics, Shenzhen University, Shenzhen, China. Her cur- 1049 rent research interests include pattern recognition, machine learning, fuzzy 1050 sets and fuzzy logic, and their related applications. 1051

Discovering the Relationship Between Generalization and Uncertainty by Incorporating Complexity of Classification

Xi-Zhao Wang, Fellow, IEEE, Ran Wang, Member, IEEE, and Chen Xu

Abstract—The generalization ability of a classifier learned from ² a training set is usually dependent on the classifier's uncertainty, 3 which is often described by the fuzziness of the classifier's outputs 4 on the training set. Since the exact dependency relation between 5 generalization and uncertainty of a classifier is quite compli-6 cated, it is difficult to clearly or explicitly express this relation in 7 general. This paper shows a specific study on this relation from 8 the viewpoint of complexity of classification by choosing extreme 9 learning machines as the classification algorithms. It concludes 10 that the generalization ability of a classifier is statistically becom-¹¹ ing better with the increase of uncertainty when the complexity 12 of the classification problem is relatively high, and the general-13 ization ability is statistically becoming worse with the increase 14 of uncertainty when the complexity is relatively low. This paper 15 tries to provide some useful guidelines for improving the gener-16 alization ability of classifiers by adjusting uncertainty based on 17 the problem complexity.

Index Terms—Complexity of classification, extreme learning
 machine, generalization, uncertainty.

20

I. INTRODUCTION

²¹ C LASSIFICATION problem, as the central part in the ²² C fields of pattern recognition and data mining, refers ²³ to a task of assigning objects to one of several predefined ²⁴ class labels. Given a set of objects, the mathematical ²⁵ model of classification problem is a discrete-valued func-²⁶ tion that maps each object to a class label. Usually, the ²⁷ process of determining the discrete-valued function from a

Manuscript received July 4, 2016; revised November 28, 2016; accepted January 11, 2017. This work was supported in part by the National Natural Science Foundation of China under Grant 61402460, Grant 61472257, Grant 61170040, and Grant 71371063, in part by the Basic Research Project of Knowledge Innovation Program in Shenzhen under Grant JCYJ20150324140036825, in part by the Guangdong Provincial Science and Technology Plan Project under Grant 2013B040403005, and in part by the HD Video Research and Development Platform for Intelligent Analysis and Processing in Guangdong Engineering Technology Research Centre of Colleges and Universities under Grant GCZX-A1409. (*Corresponding author: Ran Wang.*)

X.-Z. Wang is with the College of Computer Science and Software Engineering, Shenzhen University, Shenzhen 518060, China (e-mail: xizhaowang@ieee.org).

R. Wang and C. Xu are with the College of Mathematics and Statistics, Shenzhen University, Shenzhen 518060, China (e-mail: wangran@szu.edu.cn; xuchen@szu.edu.cn).

Color versions of one or more of the figures in this paper are available online at http://ieeexplore.ieee.org.

Digital Object Identifier 10.1109/TCYB.2017.2653223

training set is called learning while the process of using ²⁸ the determined function to classify a new object is called ²⁹ reasoning [1]–[5]. ³⁰

For a classification problem with c classes, the reason-31 ing result is generally a *c*-dimensional vector. According to 32 the output forms of the reasoning process, existing learning 33 algorithms can be classified into two categories. In one cate-34 gory, the *c*-dimensional output vector contains one component $_{35}$ of value 1 and other components of value 0. In this situa-36 tion, the class label corresponding to the component 1 will 37 be the reasoning result. This kind of algorithms are known 38 as crisp-output algorithms, such as traditional support vector 39 machine (SVM) [6]–[10], decision tree (DT) [11], [12], etc. 40 In the other category, the c-dimensional output vector con- $_{41}$ tains components of real values within the interval [0, 1]. In 42 this situation, the class label corresponding to the maximum 43 component will be the reasoning result. If the maximum is 44 attained at more than one component, a special strategy will be 45 designed to determine the final result. This kind of algorithms 46 are acknowledged as uncertain-output algorithms, such as 47 k-nearest neighbor [2], Bayesian probability model [2], back-48 propagation (BP) methods for training feed-forward neural 49 networks [13]-[16], etc. 50

Obviously, crisp-output algorithms are special cases of 51 uncertain-output algorithms. If an algorithm belongs to the 52 crisp category, then it belongs to the uncertain category, 53 however, it is not true conversely. Most crisp-output algo-54 rithms can be extended to uncertain-output algorithms, such 55 as fuzzy SVM [17], fuzzy DT [18], etc. In this paper, we 56 will intensively investigate the uncertain-output algorithms, 57 which highlight the argument that uncertainty does exist in 58 the learning and reasoning processes. 59

On the other hand, generalization of a classifier is defined ⁶⁰ as the rate of the correctly classified objects that are not in ⁶¹ the training set. It is the most important index for evaluating a ⁶² classification algorithm since the ultimate goal for developing ⁶³ a classification model is to achieve high prediction accuracy ⁶⁴ on unseen cases. Usually, the generalization of a classifier ⁶⁵ depends on multiple factors. ⁶⁶

- The mathematical model, which has a direct impact on both the training accuracy and testing accuracy.
- 2) The algorithm for training the model parameters, which ⁶⁹ is sensitive to the prediction results. ⁷⁰

2168-2267 © 2017 IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See http://www.ieee.org/publications_standards/publications/rights/index.html for more information.

129

132

71 3) *The data distribution:* In supervised learning, there is
72 a fundamental assumption that the training data has
73 the same distribution as the testing data. The learning
74 scheme that does not follow this fundamental assump-

tion is referred to as transfer learning [19], which is out

⁷⁶ of the scope of this paper.

77 Many research efforts have been made to improve the gen-78 eralization of a classifier by considering different factors. this paper, we consider a particular model parameter, 79 In 80 i.e., the uncertainty of the classifier's outputs, which has been ⁸¹ proven in [20] to have a close relationship with the gen-82 eralization of classifier. It has been shown in [20] that the ⁸³ uncertainty of the classifier's outputs has a close relationship 84 with the generalization capability. However, this relation-85 ship is difficult to express explicitly for general cases. In 86 order to further investigate this relationship, in this paper, 87 we take into account a new index, i.e., complexity of clas-⁸⁸ sification, which can be measured in different ways [21]. To 89 the best of our knowledge, this paper makes a first attempt investigate the relationship between generalization and 90 to ⁹¹ uncertainty of a classifier by incorporating the complexity of 92 classification.

In addition, choosing an appropriate classification algorithm 93 also an important issue to conduct this research. It is note-94 is 95 worthy that any uncertain-output algorithm can be used to ⁹⁶ study the relationship between generalization and uncertainty. 97 As the commonly used classification model for various prac-98 tical problems, feed-forward neural networks will be adopted. ⁹⁹ The most notable algorithm to train a feed-forward neural network is BP. Although it has been proved in [15] and [16] 100 101 that BP network has the ability to approximate any contin-102 uous function with arbitrary precision, it is often criticized 103 to have the problems of slow convergence speed and local 104 minima. In order to overcome these deficiencies, extreme lean-105 ing machine (ELM) has been proposed as a new training 106 algorithm for single-hidden layer feed-forward neural net-107 work (SLFN) [22]. Differentiating from BP that iteratively 108 tunes the weight parameters by gradient descent technique, 109 ELM randomly chooses the weight parameters between input 110 and hidden layers and analytically solves the weight parame-111 ters between hidden and output layers through Moore–Penrose ¹¹² generalized inverse [44]–[48]. Due to the extremely fast train-¹¹³ ing speed and good prediction performance, ELM has been 114 investigated intensively and extensively in the machine learn-115 ing and data mining communities [23]–[26]. Based on the 116 aforementioned advantages, we will adopt ELM as the classi-¹¹⁷ fication algorithm in this paper. The major theoretical issues 118 of ELM can be found in [27] and [28], and the applications ¹¹⁹ of ELM to different areas, such as sparse representation can 120 be found in [29] and [30].

The rest of this paper is organized as follows. Section II reviews ELMs. Section III introduces the dependency relation between generalization and uncertainty of classifiers. Section IV discusses the complexity of classification problems. Section V analyzes the relationship between generalization and uncertainty by incorporating a complexity index. Experiments review of the section VI. Finally, conclusions are given in Section VII.

II. EXTREME LEARNING MACHINE

This section will introduce ELM, which is a noniterative 130 training algorithm for SLFNs.

A. Training of ELM

A standard SLFN for classification is a discrete function ¹³³ mapping samples to class labels. Given a training set that ¹³⁴ contains *N* arbitrarily distinct samples $\mathbb{X} = \{(\mathbf{x}_i, \mathbf{t}_i)\}_{i=1}^N \subset ^{135}$ $\mathcal{R}^n \times \{0, 1\}^c$, where $\mathbf{x}_i = [x_{i1}, x_{i2}, \dots, x_{in}]$ is the *i*th training ¹³⁶ sample, $\mathbf{t}_i = [t_{i1}, t_{i2}, \dots, t_{ic}]$ is the label vector of \mathbf{x}_i , *n* is ¹³⁷ the number of features, and *c* is the number of classes. An ¹³⁸ SLFN with \tilde{N} hidden nodes and activation function $g(\mathbf{x})$ can ¹³⁹ be expressed as

$$\sum_{j=1}^{\widetilde{N}} \beta_j g(\mathbf{w}_j \cdot \mathbf{x}_i + b_j) = \mathbf{t}_i, \ i = 1, 2, \dots, N$$
(1) 141

where $\mathbf{w}_j = [w_{j1}, w_{j2}, \dots, w_{jn}]$ is the weight linking the input 142 nodes to the *j*th hidden node, b_j is the bias of the *j*th hidden 143 node, β_j is the weight linking the *j*th hidden node to the output nodes, and sigmoid function $g(x) = (1/[1 + \exp(-x)])$ is 145 selected as the activation function. 146

In ELMs, the input weights \mathbf{w}_j and biases b_j are randomly 147 chosen, and the learning can be formulated as a minimum 148 optimization problem with a regularized term 149

$$\min_{\beta} \left\{ ||\mathbf{T} - \mathbf{H}\beta||_{2}^{2} + \mu ||\beta||_{2}^{2} \right\}, \ \mu > 0$$
 (2) 150

where **H** is the hidden layer output matrix denoted as

$$\mathbf{H}(\mathbf{w}_1, \mathbf{w}_2, \dots, \mathbf{w}_{\widetilde{N}}, b_1, b_2, \dots, b_{\widetilde{N}}, \mathbf{x}_1, \mathbf{x}_2, \dots, \mathbf{x}_N)$$

$$= \begin{bmatrix} g(\mathbf{w}_1 \cdot \mathbf{x}_1 + b_1) & \cdots & g(\mathbf{w}_{\widetilde{N}} \cdot \mathbf{x}_1 + b_1) \\ \vdots & \vdots & \vdots \end{bmatrix}$$
(3)

$$\begin{bmatrix} \vdots & \ddots & \vdots \\ g(\mathbf{w}_1 \cdot \mathbf{x}_N + b_1) & \cdots & g(\mathbf{w}_{\widetilde{N}} \cdot \mathbf{x}_N + b_{\widetilde{N}}) \end{bmatrix}_{N \times \widetilde{N}}$$

and **T** is the label matrix denoted as

151

$$\mathbf{T} = \begin{bmatrix} \mathbf{t}_1 \\ \vdots \\ \mathbf{t}_N \end{bmatrix}_{N \times c}$$
(4) 155

The optimal estimation of output weights β^* can be formulated as a regularized least square problem

$$\boldsymbol{\beta}_{\widetilde{N}\times c}^* = \left(\mathbf{H}^{\mathrm{T}}\mathbf{H} + \boldsymbol{\mu}\mathbf{I}\right)^{-1}\mathbf{H}^{\mathrm{T}}\mathbf{T}$$
(5) 158

where I is the identity matrix of suitable dimension and μ is the regularizing factor.

To this end, all the parameters $\{\mathbf{w}, b, \beta\}$ in ELM have been ¹⁶¹ fixed, and the training process is finished. ¹⁶²

ELMs have been proved to have the universal approximation capabilities [31] although the training process does not include any iteration. Under the assumption of smoothness of the underlying function, the universal approximation capability of ELMs can be guaranteed by providing a sufficiently large number of hidden nodes with certain range of **w** and *b*. 168

In comparison with BP algorithm, ELMs have a much faster 169 training speed due to the noniterative mechanism. References 170 show that ELMs can finish the training process thousands of 171 times faster than BP in some scenarios, at the same time, an 172 ¹⁷³ acceptable learning accuracy is kept. The advantages and dis-¹⁷⁴ advantages of ELMs are listed in Appendix A. Furthermore, ¹⁷⁵ one can find many improved versions for ELMs. The com-¹⁷⁶ putation of weights between hidden and output layers can ¹⁷⁷ be improved through an optimization algorithm given by ¹⁷⁸ Deng *et al.* [32] in order to avoid over-fitting. Rong *et al.* [33] ¹⁷⁹ offered a pruned ELM in which the corresponding nodes ¹⁸⁰ can be removed according to the information gain to reduce ¹⁸¹ the correlation among classes in a large network structure. ¹⁸² Feng *et al.* [34] proposed an EM-ELM in which the weights ¹⁸³ are not updated when a node is added, and the algorithm ¹⁸⁴ can update the weights and adjust the network at the same ¹⁸⁵ time. Furthermore, it is found that ELMs can online deal with ¹⁸⁶ sequential data successfully [35].

187 B. Generalized Inverse and Normal Equations

In ELMs, the weights between hidden and output layers are calculated by the generalized inverse [36]. We briefly review some connections between the generalized inverse and the normal equations. Originally, the training of ELMs contains two parts. The first is to randomly assign values in a specified interval to the weights between the input and hidden layers while the second is to determine the weights between the hidden and output layers by computing the generalized inverse of the matrix **H** as $\beta^* = \mathbf{H}^{\dagger}\mathbf{T}$. It is the minimum norm and minimum least square solution of the system of linear matrix equations $\mathbf{H}\beta = \mathbf{T}$. It is easy to prove that, if the matrix **H** is of full-rank, the solution of normal equation $\mathbf{H}^{T}\mathbf{H}\beta = \mathbf{H}^{T}\mathbf{T}$ is identical to $\beta^* = \mathbf{H}^{\dagger}\mathbf{T}$.

Noting that in Section II-A, the training process of ELMs is 201 written as $\beta^* = (\mathbf{H}^T \mathbf{H} + \mu \mathbf{I})^{-1} \mathbf{H}^T \mathbf{T}$, where μ is a regularizing 202 factor. This formula is identical to $\beta^* = \mathbf{H}^{\dagger}\mathbf{T}$ if the regulariz-203 ing factor takes value zero. It is proven in [24] that the matrix 204 is of full-rank with probability 1, and therefore, we can say Η 205 ²⁰⁶ that the solution of normal equation $\mathbf{H}^{\mathrm{T}}\mathbf{H}\beta = \mathbf{H}^{\mathrm{T}}\mathbf{T}$ is available with probability 1. In fact, the regularizing factor, which 207 208 makes the solved weights as small as possible, has the effect 209 to become the matrix **H** full of rank.

Practically the number of rows is much larger than the number of columns for an input data matrix. It implies that the transformation from computing $\beta^* = \mathbf{H}^{\dagger}\mathbf{T}$ to solving the normal system of linear matrix equations $\mathbf{H}^{T}\mathbf{H}\beta = \mathbf{H}^{T}\mathbf{T}$ and save much computational load, since the order of \mathbf{H} is $N \times \tilde{N}$ but the order of $\mathbf{H}^{T}\mathbf{T}$ is $\tilde{N} \times c$, where N is the number of input samples, \tilde{N} is the number of hidden layer nodes, and c is the number of classes. A lot of numerical experiments have confirmed this saving of computational load.

220 III. DEPENDENCY RELATION BETWEEN 221 GENERALIZATION AND UNCERTAINTY 222 OF CLASSIFIERS

In this section, we will introduce the generalization and uncertainty of a classifier. The dependency relation between generalization and uncertainty is then discussed.

A. Generalization and Uncertainty

Generally speaking, the purpose of learning is to acquire ²²⁷ the knowledge hidden in the data. Knowledge representation, ²²⁸ which has been well acknowledged as a bottle-neck problem ²²⁹ in machine learning and artificial intelligence for many years, ²³⁰ does not have a general definition but has many specific forms. ²³¹ A mathematical model, such as a set of IF-THEN rules or a ²³² neural network learned from a training set, can be regarded ²³³ as a typical form of knowledge representation. The ability ²³⁴ or performance of the learned model to predict unseen cases ²³⁵ (which are not within the training set) is called generalization. ²³⁶

Let S be a finite space of samples, $F(\mathbf{x})$ be a discrete-valued ²³⁷ function defined on S, and \mathbb{X} be a subset of S. Based on values ²³⁸ of $F(\mathbf{x})$ in \mathbb{X} , an estimator function $f(\mathbf{x})$ defined on S is given ²³⁹ by using a training algorithm. The discrete-valued function ²⁴⁰ $f(\mathbf{x})$ has the same value range as $F(\mathbf{x})$. Usually we call $f(\mathbf{x})$ ²⁴¹ as a classifier trained by the algorithm on \mathbb{X} . ²⁴²

Definition 1: The generalization of classifier $f(\mathbf{x})$ is 243 defined as 244

$$G(f) = \frac{|\{\mathbf{x} : \mathbf{x} \in \mathcal{S} - \mathbb{X}, F(\mathbf{x}) = f(\mathbf{x})\}|}{|\mathcal{S} - \mathbb{X}|}$$
(6) 245

where | | denotes the number of elements in a set.

Generalization is the most important index of evaluating ²⁴⁷ a learned model. From mathematical viewpoint, the task of ²⁴⁸ learning is to find a function $f(\mathbf{x})$ through a training set ²⁴⁹ $\mathbb{X} = \{(\mathbf{x}_i, \mathbf{t}_i)\}_{i=1}^N \subset \mathcal{R}^n \times \{0, 1\}^c$ such that $f(\mathbf{x})$ can well ²⁵⁰ approximate the objective function $F(\mathbf{x})$ both at training cases ²⁵¹ and unseen cases. The difference between $F(\mathbf{x})$ and $f(\mathbf{x})$ is ²⁵² called generalization error, which can be measured from different angles. One method is to estimate an upper bound for it, the ²⁵⁴ other is to compute $R = \int_{\mathcal{S}} [F(\mathbf{x}) - f(\mathbf{x})]^2 p(\mathbf{x}) d\mathbf{x}$, where $p(\mathbf{x})$ ²⁵⁵ is the probability density function of input \mathbf{x} . Experimentally, ²⁵⁶ the generalization can be measured by the prediction accuracy ²⁵⁷ of the classifier on a testing set.

Multiple factors have critical impacts on the generalization 259 of a classifier. 260

- Model Selection: It is hard to select the most appropriate ²⁶¹ model for a given classification task. When the training ²⁶² data is fixed, the generalizations of two models might ²⁶³ be quite different. This is due to the data distribution, ²⁶⁴ i.e., a model suitable for one type of data may not be ²⁶⁵ appropriate for another type of data. ²⁶⁶
- Training Algorithm: When a model is fixed, the subsequent work is to train the model parameters based on a training set. A model with a set of trained parameters 269 has the generalization quite different from the model 270 with another set of trained parameters. 271
- 3) Representatives of Training Data: Since both the objec- 272 tive function and its approximating function are defined 273 on a space S, one problem is that the training set 274 X should be a reasonable sampling of the space S, 275 which directly relates to the fundamental assumption of 276 machine learning that the training set has an identical 277 distribution as the testing set has. 278
- 4) *Model Knowledge Parameters:* Different from the 279 parameters inside the model that are acquired directly 280

from the training process, model knowledge parame-281 ters do not explicitly appear in the model, which are 282 usually evaluated after the training process. For exam-283 ple, the uncertainty of classifier's outputs is a typical 284 model knowledge parameter. The relationship between 285 generalization and uncertainty of a classifier is initially 286 demonstrated in [20]. This paper will conduct further 287 studies on this relationship through incorporating a new 288 index, i.e., complexity of classification. 289

290 B. Fuzziness of Classifier's Outputs

In this paper, we use fuzziness to depict the uncertainty 291 292 of a classifier's outputs. The term "fuzziness," in conjunc-293 tion with the concept of fuzzy set, was first mentioned by ²⁹⁴ Zadeh [37]. He also generalized a probability measure of 295 events that cannot be described by sharply defined collection points, and suggested using entropy in information the-296 Of ry to interpret the uncertainty associated with a fuzzy event. 297 01 De Luca and Termini [38] for the first time clearly proposed 298 ²⁹⁹ three properties that a fuzziness measure should satisfy. The term fuzziness can be interchangeable with "ambiguity" in 300 some scenarios. Klir et al. [39], [40] stated that fuzziness and 301 ³⁰² ambiguity gave two cognitive uncertainty measures.

As stated in [41], the fuzziness of a fuzzy set μ can be measured by a mapping $E(\mu):F(S) \rightarrow [0, \infty]$ where F(S)of denotes the space of all fuzzy sets defined on S, satisfying the following axioms.

1) $E(\mu) = 0$ if and only if μ is a crisp set.

2) $E(\mu)$ attains its maximum value if and only if $\forall \mathbf{x} \in \mathcal{S}: \mu(\mathbf{x}) = 0.5$.

310 3) If $\mu \leq_s \sigma$, then $E(\mu) \geq E(\sigma)$, where \leq_s is defined as

$$\begin{array}{ll} \underset{11}{\underset{12}{\text{min}}} & \mu \leq_s \sigma \Leftrightarrow \min(0.5, \mu(\mathbf{x})) \geq \min(0.5, \sigma(\mathbf{x})) \\ \underset{12}{\underset{12}{\text{max}}} & \max(0.5, \mu(\mathbf{x})) \leq \max(0.5, \sigma(\mathbf{x})) \end{array}$$

313 4)
$$E(\mu) = E(\mu')$$
 when $\forall \mathbf{x} \in \mathcal{S}: \mu'(\mathbf{x}) = 1 - \mu(\mathbf{x})$

314 5) $E(\mu \cup \sigma) + E(\mu \cap \sigma) = E(\mu) + E(\sigma).$

³¹⁵ Based on these axioms, we further introduce the following ³¹⁶ definition.

Definition 2 [32]: Let $B = \{\mu_1, \mu_2, \dots, \mu_m\}$ be a fuzzy site set, the fuzziness of B can be defined as

³¹⁹
$$E(B) = -\frac{1}{m} \sum_{i=1}^{m} \left(\mu_i \log \mu_i + (1 - \mu_i) \log(1 - \mu_i) \right).$$
(7)

 $_{320}$ It is easy to verify that formula (7) indeed satisfies $_{321}$ axioms 1–5.

Given a set of samples $\mathbb{X} = \{(\mathbf{x}_i, \mathbf{t}_i)\}_{i=1}^N \subset \mathcal{R}^n \times \{0, 1\}^c$ and a well-trained classifier, a membership matrix $\mathbf{U} = [\mu_{ij}]$ are can be obtained by matching each sample to the classifier, where $\mu_{ij} = \mu_j(\mathbf{x}_i)$ denotes the membership degree of the the sample belonging to the *j*th class, where i = 1, 2, ..., Nare and j = 1, 2, ..., c. It is worth noting that each output vector may not be a probability distribution, i.e., $\mu_{ij} \in [0, 1]$, and the equality $\sum_{i=1}^{c} \mu_{ij} = 1$ does not necessarily hold. Based on Definition 2, the fuzziness of the classifier's 330 outputs for the *i*th sample can be expressed as 331

$$E(\mu_i) = -\frac{1}{c} \sum_{j=1}^{c} \left(\mu_{ij} \log \mu_{ij} + \left(1 - \mu_{ij}\right) \log(1 - \mu_{ij}) \right).$$
(8) 332

Having the above preliminaries, in the following, we propose a new concept to describe the fuzziness of a classifier's outputs on the entire training set.

Definition 3 (Fuzziness of a Classifier's Outputs): Suppose ³³⁶ that a classifier is trained from training set X. Without loss ³³⁷ of generality, X is assumed to be a sufficient sampling of ³³⁸ the entire sample space. Let $\mathbf{U} = [\mu_{ij}]_{c \times N}$ be the membership ³³⁹ matrix given by matching each training sample to the classifier, ³⁴⁰ where *c* is the number of classes and *N* is the number of ³⁴¹ samples. Then the fuzziness of the classifier's outputs can be ³⁴² defined as ³⁴³

$$E(\mathbf{U}) = -\frac{1}{cN} \sum_{i=1}^{N} \sum_{j=1}^{c} (\mu_{ij} \log \mu_{ij} + (1 - \mu_{ij}) \log(1 - \mu_{ij})).$$
⁽⁹⁾ 344

It is noted that Definition 3 uses the fuzziness of the classifier's outputs on the training set. In a more rigorous manner, it should be defined as the fuzziness of the classifier on the whole space. Unfortunately, the fuzziness of the classifier on unseen samples is unknown. According to the fundamental assumption of supervised learning that the training set is a reasonable and sufficient sampling of the entire sample space, we can use the classifier's fuzziness on the training set to approximately replace the classifier's fuzziness on the entire sample space. 354

C. Relationship Between Generalization and Fuzziness

355

Previous study [20] shows that the classifier with higher ³⁵⁶ fuzziness of outputs has a better generalization for com-³⁵⁷ plex boundary problems when the training accuracy attains ³⁵⁸ a predefined threshold. Furthermore, it demonstrates that the ³⁵⁹ outputs of boundary samples have higher fuzziness, and ³⁶⁰ samples with higher fuzziness exhibit higher risk of misclas-³⁶¹ sification. By separating samples with high fuzziness from ³⁶² samples with low fuzziness, a divide-and-conquer learning ³⁶³ algorithm based on fuzziness categorization was proposed ³⁶⁴ in [41]. It shows that the category of sample with low or high ³⁶⁵ fuzziness plays a critical role for performance improvement. ³⁶⁶ Although these studies confirm that a relationship between ³⁶⁷ fuzziness and generalization of a classifier indeed exists, it is ³⁶⁸ difficult to explicitly express this relationship in general. ³⁶⁹

In the following, we make an investigation on data set *Spam*, ³⁷⁰ which is a binary classification data set selected from UCI ³⁷¹ machine learning repository. This data set contains 4601 sam- ³⁷² ples with 57 features. We randomly split it into two parts, ³⁷³ i.e., 70% for training and 30% for testing. ELM is used to ³⁷⁴ construct a classifier, which generates four indexes, i.e., train- ³⁷⁵ ing accuracy, testing accuracy, training fuzziness, and testing ³⁷⁶ four indexes are recorded for each repetition. ³⁷⁸

We make a statistical analysis for the 100 results. First, we 379 split the interval between the minimum and maximum fuzzi- 380 ness values into ten parts with equal length and generate ten 381



Fig. 1. Dependency relation between fuzziness and accuracy for *Spam*. (a) Histogram of training fuzziness. (b) Training accuracy. (c) Histogram of testing fuzziness. (d) Testing accuracy.

³⁸² levels of fuzziness. For instance, the minimum and maximum ³⁸³ fuzziness values for testing are 0.4889 and 0.5798, respec-³⁸⁴ tively. Then, the ten fuzziness levels for testing are generated ³⁸⁵ as level 1 = [0.4889, 0.4980), level 2 = [0.4980, 0.5071), ³⁸⁶ level 3 = [0.5071, 0.5162),..., and level 10 = [0.5707, 0.5798]. ³⁸⁷ Afterwards, we make a statistic for the number of experimen-³⁸⁸ tal trials in each fuzziness level, and plot the histograms as ³⁸⁹ shown in Fig. 1(a) and (c). Finally, we get the average train-³⁹⁰ ing or testing accuracy for each fuzziness level, and plot the ³⁹¹ changing trends as shown in Fig. 1(b) and (d).

One can see from Fig. 1 that the relationship between 392 393 accuracy and fuzziness of ELM does exist for Spam. We ³⁹⁴ further calculate the Pearson correlation coefficient. As a remark, Pearson correlation reflects the statistical relation-395 396 ship between two sets of variables with a coefficient from [-1, 1]. A positive/negative coefficient represents that the 397 398 two sets of variables are positive/negative correlated, and ³⁹⁹ the absolute value represents the correlation degree. We use 400 the median to represent each fuzziness level. Taking the testing result as an example, the correlation coefficient is 401 402 calculated between fuzziness vector [0.4935, 0.5025, 0.5116, 403 0.5207, 0.5298, 0.5389, 0.5480, 0.5571, 0.5662, 0.5753] and 404 accuracy vector [0.8536, 0.8391, 0.8279, 0.8263, 0.8214, 405 0.8194, 0.8177, 0.8111, 0.8065, 0.7524]. Finally, the corre-406 lation coefficients for training and testing are calculated as -0.7145 and -0.8625, respectively. This tells that the accu-407 408 racy and fuzziness have a negative correlation for Spam, 409 i.e., a higher fuzziness will lead to a lower accuracy, and the 410 correlation degree is high.

Although the above example demonstrates that the relation-412 ship between generalization and uncertainty does exist for data 413 set *Spam*, this relationship is difficult to express explicitly for 414 general cases. In the subsequent sections, we will attempt to 415 make this relationship clear by incorporating a new index, 416 i.e., complexity of classification.

417 IV. COMPLEXITY OF CLASSIFICATION PROBLEM

Generally, a classification problem can be described as fol-419 lows. Let S be the universal space we consider, F be a discrete 420 function defined on S. For simplicity, we suppose that func-421 tion F takes values either 0 or 1, where 0 denotes one class 422 and 1 denotes the other class. Given a subset of S, denoted 423 as \mathbb{X} , which is called the training set, the values of F on 424 \mathbb{X} are known, but the values of F on $S - \mathbb{X}$ are unknown. A classification problem is to find a function f such that f can 425 well approximate F both in X and S – X. Usually, F is called 426 an objective function, f is called a classifier acquired based on 427 training set X, the approximation error on X is called training 428 error, and the approximation error on S – X represents the 429 generalization ability of F. 430

The complexity of a classification problem refers to the ⁴³¹ complexity of function F, which implies the difficulties of the ⁴³² process of finding a quality f from X. Unfortunately, there is ⁴³³ no formal definition on the complexity of a discrete function. ⁴³⁴ From references we can find a number of indexes to describe ⁴³⁵ the complexity from different angles. It is noteworthy that the ⁴³⁶ complexity of objective function is independent on the learned ⁴³⁷ classifier f. Since the objective function F is unknown in real ⁴³⁸ applications but is known on the training set X, the indexes ⁴³⁹ in describing the complexity of F can be estimated through ⁴⁴⁰ the training set X and values of F on X. In the following, we ⁴⁴¹ give several indexes to describe the complexity of F, which ⁴⁴² are mainly chosen from [21].

A. Fisher's Discriminant Ratio 444

Fisher's discriminant ratio is an old statistical index for ⁴⁴⁵ describing the difference between two populations. Suppose ⁴⁴⁶ that μ_{1j} , μ_{2j} , σ_{1j} , and σ_{2j} are the means and variances of ⁴⁴⁷ the two populations (classes) with respect to the *j*th attribute, ⁴⁴⁸ *j* = 1, ..., *n*. Then, the Fisher's discriminant ratio for the *j*th ⁴⁴⁹ attributes is defined as ⁴⁵⁰

$$\mathfrak{f}_j = \frac{\left(\mu_{1j} - \mu_{2j}\right)^2}{\sigma_{1j}^2 + \sigma_{2j}^2}.$$
 (10) 451

It is easy to see that Fisher's discriminant ratio with respect 452 to the *j*th attribute describes the distance between two classes 453 regarding this attribute. Intuitively, the longer the distance is, 454 the easier the classification problem is, the lower the complexity will be. Thus, the complexity evaluating index is 456 defined as

$$\mathfrak{Comp}_1 = \frac{1}{\max_j \{\mathfrak{f}_j\}}.$$
(11) 458

459

B. Volume of Overlap Region

A similar measure is the volume of overlap region between 460 two class conditional distributions. It depends on, for each 461 attribute, the maximum and the minimum values of each class. 462



Fig. 2. Intuitive illustration of volume of overlap region.

⁴⁶³ We denote A_j as the *j*th attribute. Then, the overlap region ⁴⁶⁴ normalized by the range of the value spanned by both classes, ⁴⁶⁵ for each attribute A_j , can be represented as

466 \mathfrak{v}_j

$${}^{467} = \frac{\mathrm{MIN}(\max(A_j, c_1), \max(A_j, c_2)) - \mathrm{MAX}(\min(A_j, c_1), \min(A_j, c_2))}{\mathrm{MAX}(\max(A_j, c_1), \max(A_j, c_2)) - \mathrm{MIN}(\min(A_j, c_1), \min(A_j, c_2))}$$

468

$$(12)$$

⁴⁶⁹ where $\max(A_j, c_1)$, $\max(A_j, c_2)$, $\min(A_j, c_1)$, and $\min(A_j, c_2)$ ⁴⁷⁰ denotes the maximum and minimum values of attribute A_j in ⁴⁷¹ the two classes, respectively. Then, the complexity evaluating ⁴⁷² index is defined as the volume of overlap region incorporating ⁴⁷³ all the attributes

474
$$\mathfrak{Comp}_2 = \prod_{j=1}^n \mathfrak{v}_j \tag{13}$$

⁴⁷⁵ where *n* is the number of attributes. An intuitive illustration of ⁴⁷⁶ volume of overlap region for a 2-D feature space is given in ⁴⁷⁷ Fig. 2. It is noted that $\mathfrak{Comp}_2 = 0$ if the value ranges of the two ⁴⁷⁸ classes do not overlap in at least one dimension. Obviously, a ⁴⁷⁹ larger value of \mathfrak{Comp}_2 represents a higher complexity of the ⁴⁸⁰ classification problem.

481 C. Intraclass/Interclass Distance Ratio

This measure first computes the Euclidean distance from each sample to its nearest neighbor within or outside the class. Assume that d_i^{intra} or d_i^{inter} is the distance between sample \mathbf{x}_i and its nearest neighbor within or outside the class, we have

$$\begin{cases} d_i^{\text{intra}} = \min_{j \neq i, y_j = y_i} d(\mathbf{x}_i, \mathbf{x}_j) \\ d_i^{\text{inter}} = \min_{j \neq i, y_j \neq y_i} d(\mathbf{x}_i, \mathbf{x}_j) \end{cases}$$
(14)

⁴⁸⁷ where y_i and y_j represent the class labels of \mathbf{x}_i and \mathbf{x}_j , respec-⁴⁸⁸ tively. Then, it takes the average of all the intraclass distances ⁴⁸⁹ and the average of all the interclass distances, and the ratio of ⁴⁹⁰ both averages is defined as the complexity of the problem

491
$$\mathfrak{Comp}_{3} = \frac{\sum_{i=1}^{N} d_{i}^{\text{intra}}}{\sum_{i=1}^{N} d_{i}^{\text{inter}}}$$
(15)

⁴⁹² where *N* is the number of samples. Similarly, a larger value ⁴⁹³ of \mathfrak{Comp}_3 represents a higher complexity of the classification ⁴⁹⁴ problem.



Fig. 3. Two normal populations.

m

problem can be found from [21].

D. Linear Separability

Linear separability was intensively discussed in the early 496 literature. A simple definition to describe the linear separability for both separable and nonseparable cases is given by 498 Smith [42] 499

in
$$\mathbf{a}^{\mathrm{T}}\mathbf{t}$$
, s.t. $\mathbf{Z}^{\mathrm{T}}\mathbf{w} = \mathbf{t} \ge \mathbf{b}$ (16) 500

where **a** and **b** are arbitrary constant vectors, **w** is the weight ⁵⁰¹ vector, $\mathbf{t} \ge 0$ is the error vector, and **Z** is a matrix in which ⁵⁰² each column **z** is defined based on the input vector **x** and its ⁵⁰³ class label **c** ⁵⁰⁴

$$\begin{aligned} \mathbf{z} &= +\mathbf{x} & \text{if } \mathbf{c} = c_1 \\ \mathbf{z} &= -\mathbf{x} & \text{if } \mathbf{c} = c_2. \end{aligned}$$
 (17) 505

The value of the objective function denotes the degree of being 506 separable for two class cases, that is 507

$$\mathfrak{Comp}_4 = \mathbf{a}^{\mathrm{T}} \mathbf{t}. \tag{18} \quad 508$$

It is noted that $\mathfrak{Comp}_4 = 0$ if the problem is linear separable. 509 Other indexes to describe the complexity of classification 510

In this section, we give an analysis on the relationship ⁵¹⁵ between generalization and uncertainty by incorporating the ⁵¹⁶ complexity of classification. Since it is difficult for us to give ⁵¹⁷ a general analysis for all the complexity indexes, we only ⁵¹⁸ adopt the index of Fisher's discriminant ratio in Section IV-A, ⁵¹⁹ and give an explanation from the viewpoint of discriminant ⁵²⁰ analysis, which has the principal of maximum probability. ⁵²¹

Without loss of generality, we consider the 1-D case, which 522 can be easily extended to multiple-dimensional cases. A normal distribution with mean μ and variance σ^2 , denoted by 524 $N(\mu, \sigma^2)$, has a probability density function 525

$$f(x) = \frac{1}{\sqrt{2\pi}\sigma} \exp\left(-\frac{(x-\mu)^2}{2\sigma^2}\right), \ -\infty < x < +\infty.$$
(19) 526

Suppose that there are two normal populations denoted by ${}_{527}$ $N(\mu_1, \sigma_1^2)$ and $N(\mu_2, \sigma_2^2)$ as shown in Fig. 3, and $x(\mu_1 < x < {}_{528}$ $\mu_2)$ is a new sample that needs to be discriminated. ${}_{529}$

For a classification problem, each population represents a $_{530}$ class. From traditional textbook [43] we can view a simple $_{531}$ way to judge sample *x* belonging to which class. $_{532}$

495

Let *C* be the cross-point between two density functions, $_{534}$ i.e., *C* satisfies the following equation:

⁵³⁵
$$\frac{1}{\sqrt{2\pi}\sigma_1} \exp\left(-\frac{(C-\mu_1)^2}{2\sigma_1^2}\right) = \frac{1}{\sqrt{2\pi}\sigma_2} \exp\left(-\frac{(C-\mu_2)^2}{2\sigma_2^2}\right).$$
⁵³⁶ (20)

⁵³⁷ It is easy to check that the cross-point locates in the interval ⁵³⁸ (μ_1, μ_2). The probabilities of sample *x* belonging to the two ⁵³⁹ classes, denoted as (α, β), can be approximately viewed as

540
$$(\alpha, \beta) = \left(\frac{1}{\sqrt{2\pi}\sigma_1} \exp\left(-\frac{(x-\mu_1)^2}{2\sigma_1^2}\right), \frac{1}{\sqrt{2\pi}\sigma_2} \exp\left(-\frac{(x-\mu_2)^2}{2\sigma_2^2}\right)\right)$$
 (21)

⁵⁴² which induces the following discriminant rules based on the ⁵⁴³ principle of maximum probability.

1) IF x < C ($\alpha > \beta$) THEN x belongs to class I.

545 2) IF x > C ($\alpha < \beta$) THEN x belongs to class II.

546 3) IF x = C ($\alpha = \beta$) THEN the class of x is uncertain.

We now relate these discussions about discriminant analysis 547 548 to the theme of this paper, i.e., uncertainty and complexity of 549 a classification problem. According to Section IV-A, the com-550 plexity of a classification problem can be described by means 551 and variances of class distributions. It can be roughly sum-552 marized as: the complexity is going up with either increasing 553 the variances (σ_1^2, σ_2^2) or decreasing the difference between 554 both means $|\mu_1 - \mu_2|$. Moreover, the uncertainty of a classifier is evaluated based on the probability vector (α , β) defined 556 in (21). According to Section III, there are many specific 557 formulas to evaluate the uncertainty (e.g., the fuzziness in 558 Definition 3), but all of them have to satisfy the conditions 559 given in Section III-B, e.g., if $\alpha < \beta$, when $\alpha' < \alpha$ and $\beta' > \beta$, 560 the uncertainty output by vector (α', β') should be smaller than that output by (α, β) . It shows that, to some extent, the differ-561 ⁵⁶² ence between the two probability values denotes the magnitude of uncertainty. The bigger the difference is, the smaller the 563 uncertainty is. Based on these analyses, we have the following 564 565 theorems.

566 Theorem 1: Let

$$_{567} g(\sigma) = \frac{1}{\sqrt{2\pi}\sigma} \left(\exp\left(-\frac{(x-\mu_2)^2}{2\sigma^2}\right) - \exp\left(-\frac{(x-\mu_1)^2}{2\sigma^2}\right) \right)$$

⁵⁶⁸ where $\sigma > 0$, $\mu_1 < \mu_2$, $x \in ([(\mu_1 + \mu_2)/2], \mu_2)$, and μ_1, μ_2 ⁵⁶⁹ are considered as constants. Then, there exists a number $\sigma_1 \in$ ⁵⁷⁰ $(0, \mu_2 - x)$ such that $g(\sigma)$ is monotonically decreasing in the ⁵⁷¹ interval $(\sigma_1, +\infty)$.

Proof: The proof of Theorem 1 is listed in Appendix B.
Theorem 2: Let

574
$$q(\delta) = \frac{1}{\sqrt{2\pi}} \left(\exp\left(-\frac{x - (\mu_2 - \delta)^2}{2}\right) - \exp\left(-\frac{x - (\mu_1 + \delta^*)^2}{2}\right) \right)$$

⁵⁷⁶ where x, μ_1 , and μ_2 are considered as constants, $\mu_1 < \mu_2$, ⁵⁷⁷ $\delta^* = |[(\mu_1 - x)/(\mu_2 - x)]|\delta$, and $\delta > 0$. Then, there exists a number δ_1 such that $q(\delta)$ is monotonically decreasing in the 578 interval $(0, \delta_1)$.

Proof: The proof of Theorem 2 can be derived similarly to 580 the proof of Theorem 1.

Theorem 3: Suppose that the conditional probability outputs of a binary classifier follow two normal distributions $N(\mu_1, \sigma^2)$ and $N(\mu_2, \sigma^2)$, respectively, where $\mu_1 < \mu_2$. Let 584

$$\alpha = -\frac{1}{\sqrt{2\pi}\sigma} \exp\left(-\frac{(x-\mu_1)^2}{2\sigma^2}\right)\beta$$

$$= 1 \qquad (x-\mu_2)^2)$$
585

$$= \frac{1}{\sqrt{2\pi\sigma}} \exp\left(-\frac{(\chi - \mu_2)}{2\sigma^2}\right)$$
 580

and

$$E(\alpha, \beta) = -\frac{1}{2}(\alpha \log \alpha + (1 - \alpha) \log(1 - \alpha))$$
580

$$+ \beta \log \beta + (1 - \beta) \log(1 - \beta)).$$
 589

Assume $\beta = K\alpha$ where $K \in (1, 1 + \epsilon)$, then $E(\alpha, \beta) = {}^{590}$ E(K) is monotonically decreasing with respect to K if 591 $K\alpha > (1/2)$.

Proof: The proof of Theorem 3 is listed in Appendix C. Noting that $g(\sigma)$ in Theorem 1 or $q(\delta)$ in Theorem 2 denotes the difference between two probability density values, which can be represented as $\beta - \alpha$ in Theorem 3. Theorem 3 directly connects this difference together with the uncertainty of the classifier's outputs given in Definition 2.

Theorem 3 shows that the uncertainty of the classifier's out- 599 puts is decreasing with the increase of the difference between 600 two density values, i.e., $\beta - \alpha$, where α and β can be con- 601 sidered as the probabilities of a sample being classified as 602 classes I and II, respectively. As a result, the conclusions in 603 Theorems 1 and 2 show that the uncertainty of a classifier's 604 outputs is becoming bigger with the increase of the complex- 605 ity of the classification problem, which is represented through 606 inflating the variance in Theorem 1 and through shrinking 607 the difference between two means in Theorem 2, respectively. 608 Since in a classification problem, the complexity is inherent 609 while the uncertainty is generated by the output of a well- 610 trained classifier which has its training and testing accuracy, 611 it is reasonable to believe that some relationships exist among 612 the accuracy, uncertainty, and complexity. 613

It is noteworthy that Theorems 1–3 cannot exactly explain ⁶¹⁴ the relationships among the three indexes, i.e., accuracy, ⁶¹⁵ uncertainty, and complexity. However, to a great extent, ⁶¹⁶ they provide solid supports to the existence of the relationships. They confirm such a fact that the classifier's uncertainty will be inevitably high if the classification problem ⁶¹⁹ is complex, no matter what classifier design algorithm is ⁶²⁰ used. This statement further implies that a high-performance ⁶²¹ classifier will have high uncertainty when the problem is ⁶²² complex. ⁶²³

VI. EMPIRICAL STUDIES 624

In this section, we will conduct some empirical studies to 625 further analyze the relationships discussed in Section V. It 626 is noteworthy the discussions in Section V were made based 627

655

679

680

681

TABLE I Selected Data Sets for Experiments

No	Data Set	# Samples	# Features	# Classes
1	Libras	369	90	15
2	Breast	699	9	2
3	SPECTF	267	44	2
4	Cancer	683	9	2
5	Chart	600	60	6
6	Cotton	356	21	6
7	CT	221	36	2
8	Dermatology	366	34	6
9	Ecoli	336	7	8
10	German	1,000	24	2
11	Glass	214	9	6
12	Haberman	306	3	2
13	Heart	270	13	2
14	Vowel	990	10	11
15	Ionosphere	351	34	2
16	Australian	690	14	2
17	Pima	768	8	2
18	Plrx	182	12	2
19	Sonar	208	60	2
20	Soybean	683	35	19
21	Bupa	345	6	2
22	Transfusion	748	4	2
23	Segment	2,310	19	7
24	Wdbc	569	30	2
25	Wpbc	198	33	2
26	Yeast	1,484	8	10
27	Zoo	101	16	7
28	Spam	4,601	57	2
29	Satellite	6,435	36	6
30	OptDigits	5,620	64	10
31	Pen	10,992	16	10

⁶²⁸ on \mathfrak{Comp}_1 , i.e., Fisher's discriminant ratio. Thus, in this sec-⁶²⁹ tion, we will also adopt \mathfrak{Comp}_1 to evaluate the complexity of ⁶³⁰ classification problems.

631 A. Selected Data Sets

The data sets used in the experiments are selected from UCI machine learning repository. The detailed information regarding these data sets are summarized in Table I. Since the complexity indexes listed in Section IV are defined for binary classification problems, we transfer each multiclass data set into binary by randomly selecting 50% classes as positive and the rest 50% classes as negative.

639 B. Experimental Design

The flowchart for training the classifier and evaluating the problem complexity is listed in Algorithm 1.

It is noteworthy that the training algorithm adopted in this section is ELM. Due to the random mechanism for weight assignment, it is easy to repeat the experiment for many times. We conduct 100 experimental trials for each data set. In each trial, 70% data are randomly selected for training, and the remaining 30% data are used for testing. Each trial will protable vide a different result, and we make statistics for fuzziness, accuracy, and complexity based on the 100 results.

The number of hidden nodes in ELM is set as 20, and sigmoid activation function is utilized. The simulations are carried out under MATLAB R2011b, which are executed on a computer with an Intel Core i7-5500U CPU@2.40 GHz, 8GB memory, and 64-bit Windows 8 system.

Algorithm	1:	Train	ELM	Classifier	and	Compute
Evaluating I	ndex	tes				

- ratio.2 *Classifier training*: train a ELM classifier based on the algorithm given in section II-A.
- 3 *Testing*: test the classifier on the testing set, compute the fuzziness (Definition 3) and generalization (testing accuracy) of the classifier.
- 4 *Complexity evaluation*: compute the complexity of the classification problem, i.e., Eq. (11).

C. Experimental Analysis

1

Similar to Section III-C, we make some statistical analyses 656 on the testing results. For each data set, ten fuzziness lev- 657 els are generated by equally dividing the interval between 658 the maximum and minimum fuzziness values. We use the 659 median to represent each fuzziness level. Then, the number 660 of experimental trials for each fuzziness level is counted, 661 and the average testing accuracy for each fuzziness level is 662 calculated. Fig. 4 demonstrates the changing trend of the test- 663 ing accuracy along with the level of fuzziness. It depicts 664 the dependency relation between testing accuracy and test- 665 ing fuzziness for the classification problems. Due to space 666 limit, we only plot the results for 12 data sets out of 31. 667 Furthermore, we calculate the Pearson correlation coefficient 668 between fuzziness vector and accuracy vector for each data 669 set. It is noteworthy there are ten fuzziness levels for each 670 data set. However, from Fig. 4, we can see that the high- 671 est fuzziness level (i.e., level ten) usually cause a sharp 672 change of the testing accuracy, which may interfere the sta- 673 tistical analysis for the overall results. Thus, we only use the 674 previous nine fuzziness values and their corresponding accu- 675 racy. The correlation coefficients r are listed in Table II. We 676 artificially set up some thresholds to justify the degree of 677 correlation. 678

1) If $0 \le |r| < 0.4$, then the correlation is low.

2) If $0.4 \le |r| < 0.7$, then the correlation is medium.

3) If $0.7 \le |r| \le 1$, then the correlation is strong.

It is observed from Table II that the generalization and fuzziness have a strong or medium correlation regarding most data sets. 682

The complexities of the problems are shown in Fig. 5, 685 which are sorted according to the order numbers (i.e., 1–31) 686 in Table I. In Fig. 5, we artificially set up a threshold such 687 that the complexity higher than the threshold is called high 6688 otherwise is called low. In this case, one can view an implicit 689 relation among the complexity, generalization, and fuzziness. 690



Fig. 4. Relationship between fuzziness and generalization of ELM classifier on different data sets. (a) Australian. (b) Chart. (c) Dermatology. (d) Segment. (e) Libras. (f) OptDigits. (g) Pen. (h) Plrx. (i) Sonar. (j) Spam. (k) SPECTF. (l) Yeast.

Data	Pearson Correlation	Data	Pearson Correlation
Set	Coefficient	Set	Coefficient
1	-0.6434	17	-0.0520†
2	-0.3522†	18	0.9743★
3	0.7838★	19	-0.6896
4	-0.7835★	20	-0.9348★
5	-0.7718★	21	0.4132
6	0.3421†	22	0.4962
7	0.3744†	23	0.8728★
8	-0.9277★	24	-0.2933†
9	-0.0579†	25	-0.5559√
10	-0.1474†	26	0.6297
11	-0.5452	27	0.3470†
12	0.5803	28	- 0.9496 ★
13	0.1768†	29	0.1455†
14	-0.9362★	30	-0.9903★
15	0.5782	31	-0.9895★
16	0 7420+		

TABLE II PEARSON CORRELATION COEFFICIENT BETWEEN OUTPUT FUZZINESS AND TESTING ACCURACY

Note: For each data set, \star represents strong correlation, $\sqrt{}$ represents medium correlation, and \dagger represents low correlation.

⁶⁹¹ The generalization of a classifier trained by ELM goes up ⁶⁹² with the increase of fuzziness if the complexity of the clas-⁶⁹³ sification problem is relatively high, while the generalization ⁶⁹⁴ of a classifier trained by ELM goes down with the increase ⁶⁹⁵ of fuzziness if the complexity of the classification problem is



Fig. 5. Complexity of the classification problems.

relatively low. For instance, it can be seen from Fig. 5 that ⁶⁹⁶ the complexity values of *Segment* (data set 23) and *Plrx* (data ⁶⁹⁷ set 18) are high, in this case, the generalizations of these two ⁶⁹⁸ data sets are becoming better with the increase of fuzziness ⁶⁹⁹ as shown in Fig. 4(d) and (h). However, the complexity val-⁷⁰⁰ ues of *OptDigits* (data set 30) and *Spam* (data set 28) are ⁷⁰¹ low, in this case, the generalizations of these two data sets are ⁷⁰² becoming worse with the increase of fuzziness as shown in ⁷⁰³ Fig. 4(f) and (j). ⁷⁰⁴

By learning the complexity of classification problems from $_{705}$ Fig. 5, we grasp some factors that are resulted from the $_{706}$



Fig. 6. Relationship between fuzziness and generalization of SVM classifier on different data sets. (a) Australian. (b) Chart. (c) Dermatology. (d) Segment. (e) Libras. (f) OptDigits. (g) Pen. (h) Plrx. (i) Sonar. (j) Spam. (k) SPECTF. (l) Yeast.

707 complexity of decision boundaries. It is obvious that there are708 some relations between them.

As we know, the complexity of a classification problem room to can be intuitively regarded as the degree of difficulty for the room problem. More specifically, it is the complexity of geometrical room that divides the sample space. In classification problem, it is room that divides the sample space. In classification problem, it is room that divides the sample space. In classification problem, it is room that divides the sample space. In classification problem, it is next to the boundary function F = 0. The ability of function room to the duration of the classifier is the generalizaroom to the fuzziness of the classifier is the uncertainty of room room for the dividing unseen samples.

⁷¹⁹ When it is easy to distinguish the classes by the boundary of ⁷²⁰ function F, it will also be easy to divide the unseen samples by ⁷²¹ f, since the structure of training data is supposed to be similar ⁷²² to the structure of unseen data and f is an estimator of F. ⁷²³ It implies that the boundary will be simple and the fuzziness ⁷²⁴ of the boundary is low. In this situation, it is reasonable to ⁷²⁵ believe that, with the decrease of classifier's fuzziness, the ⁷²⁶ generalization will be improved.

⁷²⁷ When it is difficult to distinguish the classes by the bound-⁷²⁸ ary of function F, the classifier function f is also difficult to ⁷²⁹ divide the unseen samples. It corresponds to a case of high ⁷³⁰ complexity and complex boundary. It is inherent to output ⁷³¹ high fuzziness for boundary samples for any classifier, and therefore, we reasonably believe in this situation that, with ⁷³² the increase of classifier's fuzziness, the generalization may ⁷³³ be getting better. ⁷³⁴

D. Analysis With SVM Classifiers

We further realize the above studies with SVM classifiers. ⁷⁹⁶ We adopt the "LibSVM" toolbox, the penalty term *C* is fixed ⁷³⁷ as 100, and RBF kernel $\mathcal{K}(\mathbf{x}, \mathbf{x}_i) = \exp(-[||\mathbf{x} - \mathbf{x}_i||^2/2\sigma^2])$ ⁷³⁸ with $\sigma = 1$ is adopted. The decision values of SVM are ⁷³⁹ transformed into uncertain outputs by logistic function. The ⁷⁴⁰ dependency relation between generalization and fuzziness ⁷⁴¹ regarding the 12 data sets in Fig. 4 are demonstrated in Fig. 6. ⁷⁴² It can be observed that the results are basically consistent with ⁷⁴³ those in Section VI-C, but the changing trends are not as clear ⁷⁴⁴ as those of ELM. As a result, ELM might be more suitable to ⁷⁴⁵ conduct this paper, since it has a higher degree of uncertainty ⁷⁴⁶ due to the random mechanism for input weights assignment. ⁷⁴⁷

VII. CONCLUSION 748

735

This paper finds an empirical relationship among the complexity of a classification problem, the uncertainty of classifier's outputs, and the prediction accuracy of the classifier. By experimental validation and theoretical explanation through a simple model of discriminant analysis, it is found that with the ⁷⁵⁴ increase of the uncertainty of the classifier's outputs, empiri-⁷⁵⁵ cally the accuracy is upgrading for high-complexity problem ⁷⁵⁶ but downgrading for low-complexity problem. Based on these ⁷⁵⁷ findings, in order to choose a better classification rule for a ⁷⁵⁸ practical problem, one can tune the model parameters such that ⁷⁵⁹ the uncertainty becomes larger for problems with higher com-⁷⁶⁰ plexity, or smaller for problems with lower complexity under ⁷⁶¹ the condition that an acceptable training accuracy is kept.

762 APPENDIX A 763 FEATURES OF ELMS

⁷⁶⁴ In the following, we briefly review the major advantages ⁷⁶⁵ of ELMs.

- The first advantage of ELMs is the fast training speed.
 Since the training of ELMs does not include iterative tuning, it statistically shows that ELM is thousands of times faster than BP given a predefined threshold for training accuracy.
- 2) Another feature of ELMs is the acceptable generalization ability. In comparison with other popular classification or regression algorithms, such as DTs, SVMs,
 logistic regressions, etc., the generalization of ELMs
 may not be the best in general. But so far, one cannot
 find a significant difference among the generalizations
 of these algorithms.
- The training procedure of ELMs can process online
 sequential data conveniently, which demonstrates strong
 potentials for big data analytic. It is shown that ELMs
 can effectively handle both numerical and nominal
 attributes for both classification and regression problems.

4) Mathematically it is proven that ELMs have the universal approximation ability if the activation function is differentiable. That is, ELMs can uniformly approximate any continuous function defined in an interval when the number of hidden nodes goes to infinity. This conclusion establishes the foundation of applying ELMs to various classification and regression problems.

It is worthy noting that any learning algorithm cannot be
 consistently better than others. In the following, we list several
 disadvantages of ELMs.

 As aforementioned, the weights between input and hidden layers in ELMs are randomly selected from an interval. ELMs are sensitive to this interval, and the change of the interval will produce quite different classifiers, which seriously decreases the stability.

The number of hidden layer nodes is critical for building
an ELM. A large number will lead to the generalization
decreasing but a small number can result in the training
error increasing. So far, how to select the number of
hidden layer nodes is still a challenging issue.

803Appendix B804Proof of Theorem 1

⁸⁰⁵ The original problem can be represented as

$$g(\sigma) = \frac{1}{\sqrt{2\pi}\sigma} \left(\exp\left(-\frac{(x-b)^2}{2\sigma^2}\right) - \exp\left(-\frac{(x-a)^2}{2\sigma^2}\right) \right)$$

prove that there exits σ_1 such that $g(\sigma)$ is monotonically ⁸⁰⁷ increasing when $\sigma < \sigma_1$ and $g(\sigma)$ is monotonically decreasing ⁸⁰⁸ when $\sigma > \sigma_1$.

The constant term $\sqrt{2\pi}$ can be neglected. Let $(x - a) = {}_{810}$ $k \times (b - x)$ and $\sigma = t \times (b - x)$, the original problem can be ${}_{811}$ simplified as

$$g(t) = \frac{1}{t} \left(\exp(-\frac{1}{2t^2}) - \exp(-\frac{k^2}{2t^2}) \right), \quad k > 1 \text{ and } t > 0$$
 813

prove that there exits t_1 such that g(t) is monotonically ⁸¹⁴ increasing when $t < t_1$ and g(t) is monotonically decreasing ⁸¹⁵ when $t > t_1$.

We get the first-order derivation of g(t), that is

$$g'(t) = \frac{1}{t^4} \left[\left(1 - t^2 \right) \exp\left(-\frac{1}{2t^2} \right) - \left(k^2 - t^2 \right) \exp\left(-\frac{k^2}{2t^2} \right) \right].$$
 B1B

Having this derivation, it can be derived as follows.

- 1) When t > k, $t^2 1 > t^2 k^2 > 0$ and $\exp(-[1/2t^2]) > \frac{1}{2} \exp(-[k^2/2t^2])$, thus $(t^2 1)\exp(-[1/2t^2]) > (t^2 \frac{1}{2})$ $k^2)\exp(-[k^2/2t^2])$, thus we have g'(t) < 0.
- 2) When $k \ge t > 1$, $(1 t^2) \exp(-[1/2t^2]) < 0$, thus ⁸²³ $(k^2 - t^2) \exp(-[k^2/2t^2]) > 0$, thus we have g'(t) < 0. ⁸²⁴ 3) When t = 1, we have $g'(t) = [1/t^4][-(k^2 - 825)t^2) \exp(-[k^2/2t^2])] < 0$.

So far, we have proved that g'(t) < 0 when $t \ge 1$, which B27 means that g(t) is monotonically decreasing when $t \ge 1$. B28

When 1 > t > 0 and $t \to 0$, we have $[(1 - t^2)/(k^2 - t^2)] \to 629$ $(1/k^2)$ and $\exp([(1 - k^2)/2t^2]) \to 0$ (noting that $t \le 1 < k$). 630 There exists $t^* \in (0, 1)$ such that $([(1 - t^{*2})/(k^2 - t^{*2})] > 831$ $\exp([(1 - k^2)/2t^{*2}]) = [\exp(1/2t^{*2})/\exp([k^2/2t^{*2}])]$, thus 632 $([(1 - t^{*2})\exp(-1/2t^{*2})]/[(k^2 - t^{*2})\exp(-k^2/2t^{*2})]) > 1$, 633 thus $(1 - t^{*2})\exp(-1/2t^{*2}) > (k^2 - t^{*2})\exp(-k^2/2t^{*2})$, thus 634 $g'(t^*) > 0$. 635

According to Zero theorem, there exits $t_1 \in (0, 1)$ such that ⁸³⁶ $g'(t_1) = 0$. Since g'(t) is continuous and differentiable, if all ⁸³⁷ the stagnation points are maximum points, then there is only ⁸³⁸ one stagnation point, otherwise minimum point exists. ⁸³⁹

We further get the second-order derivation of g(t), that is 840

$$g''(t) = \frac{1}{t^7} \left\{ \left[2t^2 \left(t^2 - 1 \right) - 2t^2 + (1 - t^2) \right] \exp\left(-\frac{1}{2t^2} \right) \right\}$$

$$-\left[2t^{2}\left(t^{2}-k^{2}\right)-2t^{2}k^{2}+k^{2}\left(k^{2}-t^{2}\right)\right]\exp\left(-\frac{k}{2t^{2}}\right)\right\}.$$
 842

Put the stagnation point t_1 into g''(t), since $(1 - t_1^2) = t_1^2 \exp(-1/2t_1^2) - (k^2 - t_1^2) \exp(-k^2/2t_1^2) = 0$, we have

$$g''(t_1) = \frac{1}{t_1^7} \left\{ -2t_1^2 \left[\exp\left(-\frac{1}{2t_1^2}\right) - k^2 \exp\left(-\frac{k^2}{2t_1^2}\right) \right] \right\}^{845} + \left(1 - t_1^2\right) \exp\left(-\frac{1}{2t_1^2}\right) - k^2 \left(k^2 - t_1^2\right) \exp\left(-\frac{k^2}{2t_1^2}\right) \right\}^{845}$$

+
$$\left(1 - t_{1}^{2}\right) \exp\left(-\frac{1}{2t_{1}^{2}}\right) - k^{2}\left(k^{2} - t_{1}^{2}\right) \exp\left(-\frac{k^{2}}{2t_{1}^{2}}\right)$$
. 846

Based on

$$\left(1 - t_1^2\right) \exp\left(-\frac{1}{2t_1^2}\right) - \left(k^2 - t_1^2\right) \exp\left(-\frac{k^2}{2t_1^2}\right) = 0 \qquad \text{848}$$

$$k > 1 \text{ and } 1 > t_1 > 0 \qquad \text{849}$$

817

819

850 we have

$$\sup \left(-\frac{1}{2t_1^2} \right) - k^2 \exp \left(-\frac{k^2}{2t_1^2} \right)$$
$$= t_1^2 \left[\exp \left(-\frac{1}{2t_1^2} \right) - \exp \left(-\frac{k^2}{2t_1^2} \right) \right]$$

854 and

855
$$\left(1 - t_1^2\right) \exp\left(-\frac{1}{2t_1^2}\right) - k^2 \left(k^2 - t_1^2\right) \exp\left(-\frac{k^2}{2t_1^2}\right)$$

856
$$< \left(1 - t_1^2\right) \exp\left(-\frac{1}{2t_1^2}\right) - \left(k^2 - t_1^2\right) \exp\left(-\frac{k^2}{2t_1^2}\right)$$

857
$$= 0.$$

⁸⁵⁸ Thus, $g''(t_1) < 0$, t_1 is the maximum point, which means ⁸⁵⁹ that g(t) is monotonically increasing when $t < t_1$ and g(t) is ⁸⁶⁰ monotonically decreasing when $t_1 < t < 1$.

To this end, we have proved that g(t) is monotonically increasing when $t < t_1$ and g(t) is monotonically decreasing when $t > t_1$.

864 APPENDIX C

1

Proof of Theorem 3

Substituting β with $K\alpha$ in E(K), we have

$$E(K) = -\frac{1}{2} \left(\alpha \log \alpha + (1 - \alpha) \log(1 - \alpha) + K\alpha \log(K\alpha) + (1 - K\alpha) \log(1 - K\alpha) \right)$$

Taking derivative of E(K) with respect to K, we obtain

$$\frac{dE(K)}{d(K)} = -\frac{1}{2}(\alpha \log(K\alpha) - \alpha \log(1 - K\alpha))$$

871

874

870

865

$$=-\frac{1}{2}\log\frac{1-k}{1-k}$$

⁸⁷² It is easy to view that [dE(K)/d(K)] < 0 if $K\alpha > (1/2)$, ⁸⁷³ which completes the proof.

REFERENCES

- [1] W. W. Y. Ng, A. P. F. Chan, D. S. Yeung, and E. C. C. Tsang,
 "Quantitative study on the generalization error of multiple classifier systems," in *Proc. IEEE Int. Conf. Syst. Man Cybern.*, 2005,
 pp. 889–894.
- [2] C. Bishop, *Pattern Recognition and Machine Learning*. New York, NY,
 USA: Springer-Verlag, 2006.
- [3] X. D. Wu *et al.*, "Top 10 algorithms in data mining," *Knowl. Inf. Syst.*, vol. 14, no. 1, pp. 1–37, 2008.
- [4] Z. Yan and C. Xu, "Studies on classification models using decision boundaries," in *Proc. 8th IEEE Int. Conf. Cogn. Informat.*, Hong Kong, 2009, pp. 287–296.
- [5] R. O. Duda, P. E. Hart, and D. G. Stork, *Pattern Classification*.
 New York, NY, USA: Wiley, 2012.
- [6] V. N. Vapnik, *The Nature of Statistical Learning Theory*. New York,
 NY, USA: Springer-Verlag, 2000.
- [7] R. Wang, S. Kwong, and D. Chen, "Inconsistency-based active learning for support vector machines," *Pattern Recognit.*, vol. 45, no. 10, pp. 3751–3767, 2012.
- [8] R. Wang and S. Kwong, "Active learning with multi-criteria decision making systems," *Pattern Recognit.*, vol. 47, no. 9, pp. 3106–3119, 2014.
- [9] R. Wang, D. Chen, and S. Kwong, "Fuzzy-rough-set-based active learning," *IEEE Trans. Fuzzy Syst.*, vol. 22, no. 6, pp. 1699–1704, Dec. 2014.

- [10] R. Wang, C.-Y. Chow, and S. Kwong, "Ambiguity-based multiclass as active learning," *IEEE Trans. Fuzzy Syst.*, vol. 24, no. 1, pp. 242–248, 899 Feb. 2016.
- J. R. Quinlan, "Induction of decision trees," *Mach. Learn.*, vol. 1, no. 1, 901 pp. 81–106, 1986.
- [12] R. Wang, S. Kwong, X.-Z. Wang, and Q. Jiang, "Segment based decision 903 tree induction with continuous valued attributes," *IEEE Trans. Cybern.*, 904 vol. 45, no. 7, pp. 1262–1275, Jul. 2015. 905
- M. T. Hagan and M. B. Menhaj, "Training feedforward networks with 906 the Marquardt algorithm," *IEEE Trans. Neural Netw.*, vol. 5, no. 6, 907 pp. 989–993, Nov. 1994.
- [14] G.-B. Huang, L. Chen, and C.-K. Siew, "Universal approximation using 909 incremental constructive feedforward networks with random hidden 910 nodes," *IEEE Trans. Neural Netw.*, vol. 17, no. 4, pp. 879–892, Jul. 2006. 911
- [15] K.-I. Funahashi, "On the approximate realization of continuous mappings by neural networks," *Neural Netw.*, vol. 2, no. 3, pp. 183–192, 913 1989.
- [16] G. Cybenko, "Approximation by superpositions of a sigmoidal function," 915 Math. Control Signals Syst., vol. 2, no. 4, pp. 303–314, 1989.
- [17] C.-F. Lin and S.-D. Wang, "Fuzzy support vector machines," *IEEE* 917 *Trans. Neural Netw.*, vol. 13, no. 2, pp. 464–471, Mar. 2002.
- [18] C. Z. Janikow, "Fuzzy decision trees: Issues and methods," *IEEE Trans.* 919 Syst., Man, Cybern. B, Cybern., vol. 28, no. 1, pp. 1–14, Feb. 1998. 920
- [19] Z. Deng, K.-S. Choi, Y. Jiang, and S. Wang, "Generalized hiddenmapping ridge regression, knowledge-leveraged inductive transfer learning for neural networks, fuzzy systems and kernel methods," *IEEE Trans.* 923 *Cybern.*, vol. 44, no. 12, pp. 2585–2599, Dec. 2014. 924
- [20] X.-Z. Wang *et al.*, "A study on relationship between generalization abilities and fuzziness of base classifiers in ensemble learning," *IEEE Trans.* 926 *Fuzzy Syst.*, vol. 23, no. 5, pp. 1638–1654, Oct. 2015.
- [21] T. K. Ho and M. Basu, "Complexity measures of supervised classification problems," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 24, no. 3, 929 pp. 289–300, Mar. 2002. 930
- [22] G.-B. Huang, Q.-Y. Zhu, and C.-K. Siew, "Extreme learning 931 machine: Theory and applications," *Neurocomputing*, vol. 70, nos. 1–3, 932 pp. 489–501, 2006.
- [23] E. Soria-Olivas *et al.*, "BELM: Bayesian extreme learning machine," 934 *IEEE Trans. Neural Netw.*, vol. 22, no. 3, pp. 505–509, Mar. 2011.
- [24] G.-B. Huang, H. M. Zhou, X. J. Ding, and R. Zhang, "Extreme learning 936 machine for regression and multiclass classification," *IEEE Trans. Syst.*, 937 *Man, Cybern. B, Cybern.*, vol. 42, no. 2, pp. 513–529, Apr. 2012.
- [25] A. A. Mohammed, R. Minhas, Q. M. J. Wu, and M. A. Sid- 939 Ahmed, "Human face recognition based on multidimensional PCA 940 and extreme learning machine," *Pattern Recognit.*, vol. 44, nos. 10–11, 941 pp. 2588–2597, 2011. 942
- [26] K. A. Toh, "Deterministic neural classification," *Neural Comput.*, 943 vol. 20, no. 6, pp. 1565–1595, 2008.
- [27] X. Liu, S. Lin, J. Fang, and Z. Xu, "Is extreme learning machine feasible? A theoretical assessment (part I)," *IEEE Trans. Neural Netw. Learn.* 946 *Syst.*, vol. 26, no. 1, pp. 7–20, Jan. 2015.
- [28] S. Lin, X. Liu, J. Fang, and Z. Xu, "Is extreme learning machine feasible? A theoretical assessment (part II)," *IEEE Trans. Neural Netw.* 949 *Learn. Syst.*, vol. 26, no. 1, pp. 21–34, Jan. 2015.
- [29] J. Cao, K. Zhang, M. Luo, C. Yin, and X. Lai, "Extreme learning 951 machine and adaptive sparse representation for image classification," 952 *Neural Netw.*, vol. 81, no. C, pp. 91–102, Sep. 2016. 953
- [30] J. Cao, J. Hao, X. Lai, C.-M. Vong, and M. Luo, "Ensemble extreme 954 learning machine and sparse representation classification algorithm," 955 J. Franklin Inst., vol. 353, no. 17, pp. 4526–4541, 2016.
- B. Igelnik and Y.-H. Pao, "Stochastic choice of basis functions in adaptive function approximation and the functional-link net," *IEEE Trans.* 958 *Neural Netw.*, vol. 6, no. 6, pp. 1320–1329, Nov. 1995.
- [32] W. Deng, Q. Zheng, and L. Chen, "Regularized extreme learning 960 machine," in *Proc. IEEE Symp. Comput. Intell. Data Min.*, Nashville, 961 TN, USA, 2009, pp. 389–395.
- [33] H.-J. Rong, Y.-S. Ong, A.-H. Tan, and Z. Zhu, "A fast pruned-extreme 963 learning machine for classification problem," *Neurocomputing*, vol. 72, 964 nos. 1–3, pp. 359–366, 2008. 965
- [34] G. Feng, G.-B. Huang, Q. Lin, and R. Gay, "Error minimized extreme 966 learning machine with growth of hidden nodes and incremental learning," *IEEE Trans. Neural Netw.*, vol. 20, no. 8, pp. 1352–1357, 968 Aug. 2009. 969
- [35] N.-Y. Liang, G.-B. Huang, P. Saratchandran, and N. Sundararajan, 970 "A fast and accurate online sequential learning algorithm for feedforward 971 networks," *IEEE Trans. Neural Netw.*, vol. 17, no. 6, pp. 1411–1423, 972 Nov. 2006. 973

- 974 [36] G.-B. Huang, Q.-Y. Zhu, and C.-K. Siew, "Extreme learning machine:
 975 A new learning scheme of feedforward neural networks," in *Proc. IEEE*
- Int. Joint Conf. Neural Netw., Budapest, Hungary, 2004, pp. 985–990.
 [37] L. A. Zadeh, "Probability measures of fuzzy events," J. Math. Anal.
- Appl., vol. 23, no. 2, pp. 421–427, 1968.
 [38] A. De Luca and S. Termini, "A definition of a nonprobabilistic entropy in
- the setting of fuzzy sets theory," *Inf. Control*, vol. 20, no. 4, pp. 301–312, 1972.
- ⁹⁸² [39] G. J. Klir, "Where do we stand on measures of uncertainty, ambiguity,
 ⁹⁸³ fuzziness, and the like?" *Fuzzy Set Syst.*, vol. 24, no. 2, pp. 141–160,
 ⁹⁸⁴ 1987.
- 985 [40] G. J. Klir and T. A. Folger, *Fuzzy Sets, Uncertainty and Information*.
 986 Englewood Cliffs, NJ, USA: Prentice-Hall, 1998.
- 987 [41] D. Sánchez and E. Trillas, "Measures of fuzziness under different uses of fuzzy sets," in *Advances in Computational Intelligence*989 (Communications in Computer and Information Science), vol. 298.
 990 Heidelberg, Germany: Springer, 2012, pp. 25–34.
- F. W. Smith, "Pattern classifier design by linear programming," *IEEE Trans. Comput.*, vol. C-17, no. 4, pp. 367–372, Apr. 1968.
- P33 [43] R. O. Duda, P. E. Hart, and D. G. Stork, *Pattern Classification*.
 New York, NY, USA: Wiley, 2001.
- ⁹⁹⁵ [44] S. Ding, N. Zhang, and J. Zhang, "Unsupervised extreme learning
 ⁹⁹⁶ machine with representational features," *Int. J. Mach. Learn. Cybern.*,
 ⁹⁹⁷ vol. 8, no. 2, pp. 587–595, Apr. 2017s.
- 998 [45] S. Balasundaram and D. Gupta, "On optimization based extreme learning machine in primal for regression and classification by functional iterative method," *Int. J. Mach. Learn. Cybern.*, vol. 7, no. 5, pp. 707–728, Oct. 2016.
- P. Liu, Y. Huang, L. Meng, and S. Gong, "Two-stage extreme learning machine for high-dimensional data," *Int. J. Mach. Learn. Cybern.*, vol. 7, no. 5, pp. 765–772, Oct. 2016.
- I. Zhang, S. Ding, and N. Zhang, "Incremental extreme learning machine based on deep feature embedded," *Int. J. Mach. Learn. Cybern.*, vol. 7,
- no. 1, pp. 111–120, Feb. 2016.
 A. Fu, C. Dong, and L. Wang, "An experimental study on stability and generalization of extreme learning machines," *Int. J. Mach. Learn.*
- 1010 *Cybern.*, vol. 6, no. 1, pp. 129–135, Feb. 2015.



Xi-Zhao Wang (M'03–SM'04–F'12) received the Doctoral degree in computer science from the Harbin Institute of Technology, Harbin, China, in 1998.

From 2001 to 2014, he was a Full Professor and the Dean of the College of Mathematics and Computer Science, Hebei University, Hebei, China. From 1998 to 2001, he was a Research Fellow with the Department of Computing, Hong Kong Polytechnic University, Hong Kong. Since 2014, he has been a Full Professor with the College of Computer Science and Software Engineering,

1022 Shenzhen University, Shenzhen, China. His current research interests include
1023 supervised and unsupervised learning, active learning, reinforcement learning,
1024 manifold learning, transfer learning, unstructured learning, uncertainty, fuzzy
1025 sets and systems, fuzzy measures and integrals, rough set, and learning from
1026 big data.

Dr. Wang was a recipient of many awards from the IEEE SMC Society. Dr. Wang was a recipient of many awards from the IEEE International Conference on Systems, Man, and Cybernetics (SMC) in 2005, 2007–2009, and 2012–2014, the Chair of the Technical Committee on Computational Intelligence of the IEEE SMC, and a Distinguished Lecturer of the IEEE SMC. He was the Program Co-Chair of the IEEE SMC 2009 and 2010. He is the Editor-in-Chief of the *International Journal of* Machine Learning and Cybernetics. He is also an Associate Editor of the IEEE TRANSACTIONS ON SYSTEMS, MAN, AND CYBERNETICS—PART B: Dr. CyBERNETICS, *Information Sciences Journal*, and the *International Journal for Pattern Recognition and Artificial Intelligence*.



Chen Xu received the B.Sc. and M.Sc. degrees 1052 from Xidian University, Xi'an, China, in 1986 and 1053 1989, respectively, and the Ph.D. degree from Xi'an 1054 Jiaotong University, Xi'an, in 1992. 1055

He joined Shenzhen University, Shenzhen, China, 1056 in 1992, where he is currently a Professor. From 1057 1999 to 2000, he was a Research Fellow with 1058 Kansai University, Suita, Japan, and the University 1059 of Hawaii, Honolulu, HI, USA, from 2002 to 2003. 1060 His current research interests include image process-1061 ing, intelligent computing, and wavelet analysis. 1062



Ran Wang (S'09–M'14) received the B.Eng. 1038 degree in computer science from the College 1039 of Information Science and Technology, Beijing 1040 Forestry University, Beijing, China, in 2009, and 1041 the Ph.D. degree from the Department of Computer 1042 Science, City University of Hong Kong, Hong Kong, 1043 in 2014. 1044

From 2014 to 2016, she was a Post-Doctoral 1045 Researcher with the Department of Computer 1046 Science, City University of Hong Kong. She is cur- 1047 rently an Assistant Professor with the College of 1048

Mathematics and Statistics, Shenzhen University, Shenzhen, China. Her cur- 1049 rent research interests include pattern recognition, machine learning, fuzzy 1050 sets and fuzzy logic, and their related applications. 1051